



NATIONAL COMPANY LAW TRIBUNAL AHMEDABAD (COURT NO. II)

CP (CAA) 39/AHM/2022

IN

CA (CAA) 30/AHM/2022

(Sections 230-232 and read with other applicable provisions of the Companies Act, 2013 read with The Companies (Compromises, Arrangements, and Amalgamations) Rules, 2016)

In the matter of Scheme of Arrangement
In the nature of Amalgamation

OF

ALEOR DERMACEUTICALS LIMITED

(Petitioner Company No. 1/Transferor Company)

WITH

ALEMBIC PHARMACEUTICALS LIMITED

(Petitioner Company No. 2/ Transferee Company)

Order Pronounced on:

29/08/2022

Coram:

DR. DEEPTI MUKESH,
HON'BLE MEMBER (JUDICIAL)
AJAI DAS MEHROTRA,
HON'BLE MEMBER (TECHNICAL)

MEMO OF PARTIES

In the matter of:

Aleor Dermaceuticals Limited,

(CIN: U24302GJ2016PLC092131) 5th Floor, Administrative Building Alembic Limited, Alembic Road Vadodara, Gujarat- 390003.

> ... Petitioner Company No. 1/ Transferor Company

AND

Alembic Pharmaceuticals Limited,

(CIN: L24230GJ2010PLC061123)

Alembic Road.

Vadodara, Gujarat- 390003.

... Petitioner Company No. 2/ Transferee Company

For the Petitioners

: Ms. Swati Soparkar, Advocate

For the Income Tax: Ms. Dhruvi Trivedi, Advocate

For RD

: Mr. B. Rama Ambedkar, Assistant Director

For OL

: Mr. Kamlesh Rathod

ORDER

This joint petition has been filed by the petitioner companies above named under Sections 230 and 232 of the Companies Act, 2013 (for brevity 'the Act') read with the Companies (Compromises, Arrangements and Amalgamations) Rules, 2016 seeking sanction of the scheme of



Arrangement in the nature of Amalgamation of the M/s. Aleor Dermaceuticals Limited (for brevity "Transferor Company") into M/s. Alembic Pharmaceuticals Limited (for brevity "Transferee Company").

- 2. From the records, it is seen that the first motion application seeking directions for convening/dispensing of the meeting of the Equity Shareholders, secured, unsecured creditors and debenture holder of the Petitioner Companies was filed before this Tribunal vide CA (CAA) 39/AHM/2022 and vide order dated 16.02.2022, application was allowed.
- 3. The 2nd motion joint petition, was admitted vide order dated 27.06.2022, issuing Notices to the concerned Regulatory authorities and also directing publication in English daily "Indian Express" Vadodara Edition and Gujarati daily "Sandesh" Vadodara Edition. The chairperson of meetings of secured creditors and unsecured creditors of transferor company and meetings of secured creditors, unsecured creditors and equity shareholders of the transferee company have filed separate report of compliance, along with affidavit dated 17.06.2022.
- 4. The Regional Director, North Western Region, MCA has filed common report on 07.07.2022 along with the report of RoC, Ahmedabad. They have sought following directions:



- The transferee company to undertake the compliance of section 232(3)(i) of the companies Act, 2013 and to pay fees, accordingly.
- ii) The transferee company is listed with BSE and NSE. In this regard the petitioner company has stated that the NOCs from the stock exchanges are not required in the matter since the scheme involves amalgamation of a wholly owned subsidiary with its parent company, as per SEBI circular master circular No. SEBI/HO/CFD/DIL1/CIR/P/2021/0000000665 on scheme of arrangement by listed entities dated 23.11.2021. The transferee company have also sent the notice u/s 230(5) to both stock exchanges on 18.05.2022.
- iii) As per the last MGT-7/ Annual Return for the Financial year 2020-21 filed under the MCA21 portal, the transferee company has 60% equity shareholding in the transferor company. However, the applicant companies have mentioned in the present scheme that transferee company does have entire equity share capital of transferor company. In this regard, Hon'ble NCLT may kindly direct the applicant companies to place the fact on record and submit adequate supporting documents in this regard.
- iv) Transferor company to place the fact on record and be complied with the provisions of section 77 of the Companies Act, 2013 read with rule 3 of the Companies (Register of Charges) Rules, 2013.



- v) The transferee company to place the fact on record and issue necessary direction to comply with requirement of section 82 of the Companies Act, 2013 read with rule 8 of the Companies (Registration of Charges) Rules, 2014.
- vi) The petitioner companies to pay legal fees/cost to the central government.
- 5. The Official Liquidator, attached with the Hon'ble High Court of Gujarat has filed observations, seeking directions as follows:
 - i) The transferor company to preserve its books of accounts, papers and records and shall not dispose them without prior permission of the Central Govt. as per the provisions of the section 239 of the Companies Act, 2013.
 - ii) The transferor company to ensure statutory compliance of all applicable laws and also on sanctioning of the scheme, the transferor company shall not be absolved from any of its statutory liability, in any manner.
 - iii) The transferee company to pay such cost and expenses to the office of the official liquidator for the transferor company or any such amount as may be considered appropriate by this Bench.
 - iv) The petitioner companies to lodge a certified copy of order along with the Scheme, with the concerned superintendent of stamp duty for determination of stamp duty payable, if any.



- 6. The Ld. Counsel for the Income Tax Department submitted that a demand of Rs. 4461/- in respect of order u/s 220(2) of the Income Tax Act is outstanding against the transferee company and no demand is outstanding against the transferor company. The department has no objection regarding the scheme.
- 7. The Petitioner Companies have filed a common affidavit in reply on 06.07.2022, to the observations of Regional Director, RoC, Official Liquidator and Income Tax Department. The submission made in above stated replies are as follows:
 - For differential fees, the transferee company gave undertaking for compliance of section 232(3)(i) of the Companies Act, 2013.
 - ii) A share purchase agreement was executed between the parties on 28.03.2022 and accordingly, all the shares held by Orbicular Pharmaceutical Technologies Private Limited in Aleor Dermaceuticals Limited were acquired by the transferee company on 28.03.2022. The consideration for the said acquisition was paid on 29.03.2022 and requisite disclosure in form of the intimation of acquisition was made to the concerned stock exchanges on 29.03.2022. A copy of the bank confirmation and a copy of the intimation of the acquisition for the said transaction to the concerned stock exchange are annexed with reply. The present scheme is proposed between the wholly owned subsidiary and the parent holding company.



It is submitted that the Petitioner Transferor Company has not sought any borrowings from the Axis Trustee Services Limited. The said institution has the Debenture Trustee Agreement with regard to the Non -Convertible Redeemable Debentures issued by Aleor Dermaceuticals Limited which are fully held by Alembic Pharmaceuticals Limited and Axis Trustee Services Limited is acting only as a Debenture Trustee and the charge is created for this limited purpose only. In view of the same, Axis Trustee Services Limited is not a Secured Creditor of the Petitioner Transferor Company. Further, the requisite charge is registered in this regard with the MCA vide Charge id- 00068198 and hence there is no violation with regard to failure to create charge as per the requirements of Section 77 of the Companies Act, 2013 read with Rule 3 of Companies (Register of Charges) Rules, 2014. It is submitted that Axis Trustee Services Limited is not a Secured Creditor but acts as the Trustee of the consortium of five Secured Creditors who have approved the proposed Scheme at a duly convened meeting. Therefore, though the Petitioner Transferee Company has five Secured Creditors, on the record of Registrar of Companies, name of their Trustee alone is reflected for the open secured charge. Out of the five Secured Creditors, four Secured Creditors had remained present at their duly convened meeting and all five Secured Creditors voted in favour of the Scheme through remote e-voting. Further, all the Secured Creditors have also



provided the No Objection Certificates for the proposed Scheme. Similar NOC dated 29th June, 2022 is also obtained from Axis Trustee Services Limited and all the NOCs are annexed with the reply.

A copy of the Security Trusteeship Agreement is annexed; which clearly reflects that on behalf of the Secured Creditors of the Petitioner Transferee Company, Axis Trustee Services Limited is acting as the Security Trustee.

iii) It is submitted that Bank of Baroda is not a Secured Creditor as there are no borrowings or other facility obtained from the said Bank. The charge is reflected in relation to an old Bank Guarantee issued by the said Bank. Against the said Bank Guarantee a Fixed Deposit was placed as 100% margin money with the said Bank. The Bank Guarantee was discharged and the amount of fixed deposit was released by Bank of Baroda on 27/04/2022. It is further submitted that the Petitioner Transferee Company has received No-objection certificate dated 29th June 2022 from Bank of Baroda for satisfaction of charge. The Petitioner Transferee Company has filed the satisfaction of charge in Form CHG-4 with MCA vide SRN F12235123 dated 30th June, 2022. Copy of said Form and updated index of charges extracted from the website of MCA is annexed.



iv) The petitioner transferee company undertakes to pay legal fees, as per the order.

- v) The transferee company gave undertaking to preserve the books of accounts, papers and records of the transferor companies and not to dispose of the same without prior permission of the central govt. as per the provisions of section 239 of the Companies Act, 2013.
- The petitioner companies have filed their respective audited financial statements as on 31st March 2022, which are annexed with the petition.
- 9. As recorded in the order dated 03.08.2022, learned counsel for the Income Tax Department, Mr. Kamlesh Rathod from the Official Liquidator Department and learned Assistant Director Mr. B. Rama Ambedkar from the Regional Director's office stated that the respective departments do not have any objections, if the scheme is approved.
- 10. Heard submissions and perused documents placed on record. Considering the approval accorded by the members and creditors of the Petitioner Companies to the proposed Scheme, and the no objection given by the respective regulatory authorities, it appears that the requirements of the provisions of section 230 and 232 are satisfied by the petitioner companies. It seems that the proposed Scheme of Amalgamation is bona fide and in the interest of the shareholders and creditors. Therefore, petition is allowed and the Scheme envisaging amalgamation of Aleor Dermaceuticals Limited, the Petitioner Transferor Company with Alembic Pharmaceuticals Limited, the Transferee Company is hereby sanctioned. It is declared that the said sanctioned scheme shall be binding



- on the petitioner companies and their shareholders, creditors and all concerned under the scheme.
- 11. Notwithstanding the above, if any deficiency is found or, violation committed qua any enactment, statutory rule or regulation, the sanction granted by this Tribunal to the scheme will not come in the way of action being taken, albeit, in accordance with law, against the concerned persons, directors and officials of the Petitioners.
- 12. While approving the Scheme as above, we further clarify that this order should not be construed as an order in a manner granting exemption from payment of stamp duty, or taxes including income tax, GST etc., or any other charges or payment in accordance with law, or any kind of waiver in respect of any permission / compliance with any other requirement which may be specifically required under any law.

13. The Scheme is sanctioned with following directions:

- i. The Scheme of Arrangement as annexed herewith as "Annexure A" is hereby sanctioned and it is declared that the same shall be binding on the Petitioner Companies and their Shareholders and Creditors and all concerned under the Scheme.
- ii. It is declared that the Transferor Company shall be dissolved without winding up.
- iii. All the properties mentioned in schedule as "Annexure B", right and powers of the Transferor Company specified in the schedule hereto and all the other



property, rights and powers of the Transferor Company be transferred without any further act or deed to the Transferee Company and accordingly the same shall, pursuant to Section 232 of the Act, vest in the Transferee Company for all the estate and interest of the Transferor Company, therein but subject nevertheless to all charges now affecting the same, if any.

- iv. The Transferee Company (along with its nominees) holds 100% of the issued, subscribed and paid-up share capital of the Transferor Company. Accordingly, the Transferor Company is the wholly owned subsidiary of the Transferee Company. Upon the Scheme becoming effective, no shares of the Transferee Company shall be allotted in lieu or exchange of the holding of the Transferee Company in the Transferor Company and the share capital of the Transferor Company shall stand cancelled on the Effective Date without any further act, instrument, or deed. Further, upon the Scheme becoming effective, share warrants and Non-Convertible Debentures issued by the Transferor Company to the Transferee Company shall stand cancelled on the Effective Date without any further act, instrument, or deed.
- v. All proceedings, if any, now pending by or against the Transferor Company be continued by or against the Transferee Company.
- vi. All workers / employees of all the Transferor Companies shall be deemed to

 become the workers /employees of the Transferee Company with effect from

 the Appointed Date, and shall stand absorbed in the Transferee Company in
 accordance with the Scheme without any interruption of service and on terms

- and conditions no less favourable than those on which they are engaged by the Transferor Companies, as on the Effective Date.
- vii. The Petitioner companies within thirty days of the date of receipt of this order, cause a certified copy of this order to be delivered to the Registrar of Companies for registration and on such certified copy being so delivered, the entire Undertaking of the Transferor Company shall stand transferred to the Transferee Company and the Registrar of Companies shall place all documents relating to the Transferor Company to the file kept by him in relation to the Transferee Company and the files relating to the said two companies shall be treated accordingly.
- viii. All concerned Authorities to act on copy of this order along with the Scheme authenticated by the Registrar of this Tribunal shall issue the certified copy of this order along with the Scheme immediately.
 - ix. The Petitioner Companies are directed to lodge a copy of this Order and the approved Scheme and Schedule of Assets of the Transferor Company attached as Annexure-B with this order, duly authenticated by the Registrar of this Tribunal, with the concerned Superintendent of Stamps, for the purpose of adjudication of stamp duty, if any, within 60 days from the date of the Order.
 - x. The Petitioner Companies are further directed to file a copy of this order along with the copy of the Scheme with the concerned Registrar of Companies, electronically, along with e-form INC-28 in addition to physical



- copy within 30 days from the date of issuance of the certified copy of the Order by the Registry as per relevant provisions of the Act.
- xi. The legal fees and expenses of the office of the Regional Director are quantified at Rs.10,000/-in respect of both the petitioner companies. The said fees to the Regional Director shall be paid by the Transferee Company.
- xii. The legal fees and expenses of the office of the Official Liquidator are quantified at Rs.10,000/- in respect of the petitioner Transferor Company. The said fees to the Official Liquidator shall be paid by the Transferee Company.
- xiii. Any person aggrieved shall be at liberty to apply to the Tribunal in the above matter for any direction that may be necessary.
- xiv. The Company Petition CP(CAA) No. 39 of 2022 connected with CA(CAA) No. 30 of 2022 are disposed of, in terms of above order.

-Sd-AJAI DAS MEHROTRA MEMBER (TECHNICAL) -Sd-DR. DEEPTI MUKESH MEMBER (JUDICIAL)



Abhishek Singh

Annexure- "A"

ANNEXURE - F

SCHEME OF ARRANGEMENT

IN THE NATURE OF AMALGAMATION

BETWEEN

ALEOR DERMACEUTICALS LIMITED ('TRANSFEROR COMPANY')

AND

ALEMBIC PHARMACEUTICALS LIMITED ('TRANSFEREE COMPANY')

AND

THEIR RESPECTIVE SHAREHOLDERS UNDER SECTIONS 230 TO 232 AND OTHER APPLICABLE PROVISIONS OF THE COMPANIES ACT, 2013 AND RULES FRAMED THEREUNDER







This Scheme of Arrangement ("Scheme", as more particularly defined below) is presented under Sections 230 to 232 and other applicable provisions of the Act (as defined below) and Section 2(1B) and other applicable provisions of the Income-tax Act, 1961, between Alcor Dermaceuticals Limited and Alembic Pharmaceuticals Limited and their respective shareholders.

This Scheme provides for the following -

- a) Amalgamation of Aleor Dermaceuticals Limited with Alembic Pharmaceuticals Limited with effect from the Appointed Date on the agreed terms and conditions as set out herein; and
- Various other matters consequential, supplemental and/ or otherwise integrally connected therewith.









(A) BACKGROUND OF THE COMPANIES

- 1. Aleor Dermaceuticals Limited (hereinafter referred to as 'Aleor' or 'the Transferor Company') is a public limited company, having Corporate Identification Number: U24302GJ2016PLC092131, incorporated on May 23, 2016, under the provisions of Companies Act, 2013. The registered office of Aleor is situated at 5th Floor, Administrative Building Alembic Limited, Alembic Road, Vadodara 390003, India in the state of Gujarat. Aleor is engaged in the business of developing, manufacturing and selling of pharmaceutical (dermatology) products for global markets. The Transferor Company is a wholly owned subsidiary of Transferee Company.
- Alembic Pharmaceuticals Limited (hereinafter referred to as 'Alembic' or the 'Transferee Company') is a public limited company and having its equity shares listed on BSE Limited and National Stock Exchange of India Limited, having Identification Corporate Number: L24230GJ2010PLC061123, incorporated on June 16, 2010, under the provisions of Companies Act, 1956. The registered office of Alembic is situated at Alembic Road, Vadodara - 390003, India in the state of Gujarat. Alembic is an integrated pharmaceutical company principally engaged in the development, manufacturing and marketing of pharmaceuticals products i.e. Active Pharmaceuticals Ingredients and Formulations.



The proposed amalgamation between the Transferor Company and the Transferee Company shall result in the following benefits, amongst others, to both the Companies and their respective shareholders:

a) The amalgamation will enable the Transferee Company to integrate its business operations and provide impetus to the operations of the Transferee Company. The consolidation of the activities by way of an amalgamation will provide seamless access to the assets of the Transferor Company as well as technical expertise and financial strength of the Transferee Company to the new business vertical. This will lead to synergies in operations, reduction in overheads including administrative, managerial and other expenditure, operational rationalization, organizational efficiency, competitive





52

advantage and optimal utilization of resources eventually enhancing the growth and reputation of the group.

- b) The combined entity will have a bigger portfolio of products and direct access to markets, which will strengthen its capabilities to serve more efficiently its customers. This will also enable the Transferee Company to consolidate its offering of branded products to its existing customer base across different therapies for various markets, both overseas as well as domestic where it has already a strong presence with requisite resources, address the competitive regulatory environment, risks and policies, better management of supply chain, better product profiling, greater differentiation, ability to strategize the business for long term growth, consolidation and creation of shareholder value. Further the growth of business of branded products envisages adequate capital and resources commitments and hence the merger of the Transferor Company into the Transferee Company shall enable the pooling of abundant resources of the Transferee Company to the business of the Transferor Company and impetus to the growth at a consolidated level.
- Company and the Transferee Company as they are engaged in the similar business activities including common geographies, which is expected to result in cost saving, operational efficiency on account of scale, efficient and optimum resource utilization and thereby enhancing the business value for the Transferee Company through faster and effective decision making and avoiding duplication of efforts. It is believed that the faster decision making would be in the best interests of the shareholders, employees and other stakeholders.
- d) Reduction in the multiplicity of legal and regulatory compliances required at present to be carried out by both the Transferor Company and the Transferee Company.
 - The Boards of Transferor Company as well as Transferee Company believe that this merger will contribute to smooth integration of both the Companies and would benefit the shareholders, employees and





other stakeholders of the Transferor Company and the Transferee Company.

In view of the aforesaid advantages, the Board of Directors of Transferor Company and Transferee Company have considered and proposed the Scheme under provisions of Sections 230 to 232 and other applicable provisions of the Act and the rules framed thereunder.

(C) OPERATION OF THE SCHEME

This Scheme provides for amalgamation of the Transferor Company on a going concern basis with the Transferee Company in the manner set out in this Scheme and other applicable provisions of Applicable Law.

(D) TREATMENT OF THE SCHEME FOR THE PURPOSE OF THE INCOME-TAX ACT, 1961

This Scheme has been drawn up to comply with the conditions relating to "Amalgamation" as specified under Section 2(1B) of the Income-tax Act, 1961. If any of the terms or provisions of this Scheme is/ are found or interpreted to be inconsistent with the provisions of Section 2(1B) of the Income-tax Act, 1961 at a later date including resulting from an amendment of law or for any other reason whatsoever, the provisions of Section 2(1B) of the Income-tax Act, 1961 shall prevail and the Scheme shall stand modified to the extent determined necessary to comply with the provisions of Section 2(1B) of the Income-tax Act, 1961.

(E) PARTS OF THE SCHEME

This Scheme is divided into the following parts:

- a) Part I deals with the background of the Companies, overview of businesses of the Companies and the rationale of the Scheme:
- Part II deals with the definitions; date of taking effect and operative date and the share capital of the Companies;
- c) Part III deals inter alia with the amalgamation of Transferor Company with Transferee Company pursuant to Sections 230 to 232 and other applicable provisions of the Act and Section 2(1B) and other applicable provisions of the Income-tax Act, 1961; and
- d) Part IV deals with the general terms and conditions applicable to this Scheme and other matters consequential and integrally connected thereto.









1. DEFINITIONS

In this Scheme, the following words and expressions shall, unless the context requires otherwise, have the following meanings ascribed to them:

- 1.1 "Act" means the Companies Act, 2013 and rules made there under and the circulars, guidelines, notifications and orders issued by Ministry of Corporate Affairs as may be applicable, including any statutory modification, re-enactments or amendments thereof for the time being in force;
- 1.2 "Applicable Laws" mean any statute, notification, bye laws, rules, regulations, guidelines, rule of common law, policy, code, directives, ordinance, orders or instructions having the force of law enacted or issued by any Appropriate Authority including any statutory modification or re-enactment thereof for the time being in force in India;
- 1.3 "Appointed Date" means 1st April, 2021 or such other date as may be approved by the National Company Law Tribunal or any other Appropriate Authority;
- 1.4 "Appropriate Authority" or "Governmental Authority" means and includes any applicable Central, State or Local Government, legislative body, regulatory or administrative authority, Registrar of Companies, Regional Director, Official Liquidator, agency or commission or any court, tribunal, board, bureau or instrumentality thereof or arbitration or arbitral body having jurisdiction on behalf of the Republic of India or any state or province or other political subdivision thereof or any municipality, district or other subdivision thereof over Transferor Company or Transferee Company, as the context may require;



1.5 "Board of Directors" or "Board" means the respective Board of Directors of each of the Companies under the Scheme and shall include any committee or sub-committee of Directors thereof constituted or appointed and authorized for the purposes of matters pertaining to this Scheme and or any other matter relating thereto;







- 1.6 "Companies" means collectively the Transferor Company and the Transferee Company;
- 1.7 "Effective Date" means the last of the date on which the conditions specified in Clause 18 of this Scheme are fulfilled with respect to the Scheme. References in this Scheme to the "date of coming into effect of this Scheme" or "upon the Scheme being effective" shall mean the Effective Date;
- 1.8 "Encumbrance" means (i) any charge, lien (statutory or other), or mortgage, any easement, encroachment, right of way, right of first refusal or other encumbrance or security interest securing any obligation of any person; (ii) pre-emption right, option, right to acquire, right to set off or other third party right or claim of any kind, including any restriction on use, voting, selling, assigning, pledging, hypothecating, or creating a security interest in, place in trust (voting or otherwise), receipt of income or exercise; or (iii) any equity, assignments hypothecation, title retention, restriction, power of sale or other type of preferential arrangements; or (iv) any agreement to create any of the above; and the term "Encumber" shall be construed accordingly;
- 1.9 "National Company Law Tribunal" or "NCLT" means the National Company Law Tribunal, Ahmedabad bench, having jurisdiction in relation to the Transferor Company and the Transferee Company and shall be deemed to include, if applicable, a reference to such other forum or authority which may be vested with any of the powers of NCLT to sanction the Scheme under the Act;
- 1.10 "Registrar of Companies" means Registrar of Companies, Ahmedabad;
- 1.11 "Scheme" or "the Scheme" or "this Scheme" means this Scheme of Arrangement among the Companies and their respective shareholders, in accordance with the provisions hereof and pursuant to the provisions of Sections 230-232 and other applicable provisions of the Act;









- 1.12 "Transferee Company" or "Alembic" means Alembic Pharmaceuticals Limited, a listed public limited company, incorporated under the provisions of the Companies Act, 1956 and having its registered office at Alembic Road, Vadodara - 390003, India in the state of Gujarat;
- 1.13 "Transferor Company" or "Aleor" means Aleor Dermaccuticals Limited, a public limited company, incorporated under the provisions of the Companies Act, 2013 and having its registered office at 5th floor, Administrative Building, Alembic Limited, Alembic Road, Vadodara -390003, India in the state of Gujarat; and
- 1.14 "Undertaking" shall mean the operations and activities and the whole of the undertaking of the Transferor Company as a going concern, all their assets, rights, licenses, and powers, and all their debts, outstandings, liabilities, duties, obligations and employees as on the Appointed Date including, but not in any way limited to the following:
 - a) All the assets and properties of the Transferor Company, whether movable or immovable, tangible or intangible, whether corporeal or incorporeal including, without limitation, offices, buildings, and structures, research and development units, capital work-inprogress, freehold, leasehold and any other title, interests or right in such immovable assets (including land and building being held as stock in trade), furniture, fixtures, office equipment, computers, plant and machineries, equipment, interests, appliances, tools, accessories, installations, intangible assets under development, vehicles, advances, deposits, cash and bank balances and all stocks as on the Appointed Date;
 - All investments (including shares, scripts, stocks, bonds, debentures, debenture stock, units of mutual funds and other securities), if any, including dividends declared or interest accrued thereon of the Transferor Company;
 - c) All the debts, liabilities, duties and obligations of the Transferor Company as on the Appointed Date both present and future, whether provided for or not in the books of accounts or disclosed in the balance sheet, whether secured or unsecured, all guarantees, assurances, commitments and obligations of any kind, nature or description, whether fixed, contingent or absolute, asserted or unasserted, matured or unmatured, liquidated or unliquidated;









SI

accrued or not accrued, known or unknown, due or to become due, whenever or however arising pertaining to the Transferor Company; and

- d) Without prejudice to the generality of sub-clauses (a), (b) and (c) above, the Undertaking of the Transferor Company shall include:
 - including lease-hold rights, tenancy rights, industrial and other licenses, registrations, permits, authorisations, trademarks, patents and other industrial and intellectual properties, electrical connections, telephones, telex, facsimile and other communication facilities and equipment, rights and benefits of all agreements, pending applications and all other interests, rights and powers of every kind, nature and description whatsoever, privileges, liberties, easements, advantages, benefits and approvals of the Transferor Company;
 - ii). All assets, as are moveable in nature, whether present or future or contingent, tangible or intangible, in possession or not, corporeal or incorporeal, in each case, wherever situated (including plant and machinery, research and developments units, capital work in progress, furniture, fixtures, fixed assets, computers, air conditioners, appliances, accessories. office equipment, installations, vehicles, inventories, stock in trade, stores and spares, packing material, raw material. formulation, tablets, capsules, active pharmaceutical ingredients, drug intermediaries, tools and plants), actionable claims, debit notes, earnest monies and sundry debtors, prepaid expenses, bills of exchange, promissory notes. financial assets, investment and shares in entities / branches in India or overseas, outstanding loans and advances, recoverable in cash or kind or for value to be received, receivables, funds, cash and bank balances and deposits including accrued interest thereto with Government, semi-Government, local and other Appropriate Authorities and bodies, banks, customers and other persons, the benefits of any bank guarantees, performance guarantees and tax related assets and credits, including but not limited to service tax input credits, CENVAT credits, value added / sales tax / entry tax credits or set-offs, advance tax, tax deducted at source, IAX







58

collected at source, self assessment tax, right to carry forward and set-off accumulated losses and unabsorbed depreciation, right to carry forward and set-off book losses (including unabsorbed depreciation), set-off or credit of Minimum Alternate Tax, Goods and Services tax (GST), Equalization levy, Dividend Distribution Tax, and other indirect taxes and tax refunds:

- iii). All permits, licenses, permissions, approvals including but not limited to Abbreviated New Drug Applications ('ANDAs'), tentative and/or final, clearances, consents, benefits, registrations, rights, entitlements, credits, certificates, awards, sanctions, allotments, quotas, no objection certificates, exemptions, concessions, subsidies, incentives, tax deferrals and exemptions and other benefits (in each case including the benefit of any applications made for the same), income tax benefits, deductions and exemptions, liberties and advantages, approval for commissioning of project and other licenses or clearances, granted / issued / given by any Appropriate Authorities, organizations or companies including those relating to privileges, powers, facilities of every kind and description of whatsoever nature and the benefits thereto;
- iv). All application monies, advance monies, earnest monies and/or security deposits paid or deemed to have been paid and payments against other entitlements of the Transferor Company;
- v). All books, records, files, papers, engineering and process information, software licenses (whether proprietary or otherwise), test reports, computer programs, drawings, manual, data, databases including databases for procurement, commercial and management, catalogues, quotations, sales and advertising materials, product registrations, dossiers, product master cards, list of present and former customers and suppliers including service providers, other customer information, customer credit information, customer / supplier pricing information, and all other books and records, whether in physical or electronic form;
- vi). All intellectual property rights (including applications for registrations of the same and the right to use such intellectual property rights), applications, trade names, service names and



marks, patents, copyrights, project design, market authorization, approvals, marketing intangibles, special status, trade secrets, research and studies, technical know-how, confidential information, domain names, books, records, files, papers, software licenses (whether proprietary or otherwise), and other intellectual property rights of any nature whatsoever and data and all other records and documents, whether in physical or electronic form relating to the activities and operations of the Transferor Company;

-

- vii). Amounts claimed by the Transferor Company whether or not recorded in the books of account of the Transferor Company from any person including Governmental Authority, under any law, act or rule in force, as refund of any tax, duty, cess or of any excess payment;
- viii). Approvals of whatsoever nature (including but not limited to benefits of all tax holiday, tax relief / benefit available/ to be available to the eligible units, exemption / deduction under Chapter III of Income-tax Act, 1961, deduction under Chapter VI-A of Income-tax Act, 1961 for the eligible period) and wheresoever situated, belonging to or in the ownership, power or possession or control of or vested in or granted in favour of or enjoyed by the Transferor Company as on the Appointed Date or any other benefits and advantages of whatsoever nature and wheresoever situated belonging to or in the ownership, power, use or possession and in the control of or vested in or granted in favour of or enjoyed by the Transferor Company;
- ix). Right to any claim not preferred or made by the Transferor Company in respect of any refund of tax, duty, cess or other charge, including any erroneous or excess payment thereof made by the Transferor Company and any interest thereon, with regard to any law, act or rule or scheme made by the Governmental Authority, and in respect of set-off, carry forward of unabsorbed losses, deferred revenue expenditure, deduction, exemption, rebate, allowance, amortization benefit, etc. under the Income-tax Act, 1961, or taxation laws of other countries, or any other or like benefits under the said acts or under and in accordance with any law or act, whether in India or anywhere outside India;







- x). All rights to use and avail telephones, facsimile, email, internet, leased line connections and installations, utilities, electricity and other services, reserves, provisions, funds, benefits of assets or properties or other interests held in trusts, registrations, engagements, arrangements of all kind, privileges and all other rights, easements, liberties and advantages of whatsoever nature and wheresoever situated belonging to or in the ownership, power or possession and in control of or vested in or granted in favour of or enjoyed by the Transferor Company and all other interests of whatsoever nature belonging to or in the ownership, power, possession or control of or vested in or granted in favour of or held for the benefit of or enjoyed by the Transferor Company;
- xi). All rights, contracts, agreements, engagements, guarantees. purchase orders / service orders, operation and maintenance contracts, memoranda of understandings, memoranda of agreements, memoranda of agreed points, bids, tenders, tariff policies, expressions of interest, letters of intent, hire and purchase arrangements, power purchase agreements, lease / license agreements, tenancy rights, joint development agreements, agreements / panchnamas for right of way, equipment purchase agreements, agreement with customers. purchase and other agreements with the supplier ,' manufacturer of goods service providers. arrangements, undertakings, deeds, bonds, schemes, concession agreements, insurance covers and claims, clearances and other instruments of whatsoever nature and description, whether vested or potential and written, oral or otherwise and all rights, title, interests, claims and benefits thereunder forming part of the Transferor Company;
- xii). All liabilities including all debts (whether in Indian Rupees or foreign currency), loans raised and used, obligations incurred, whether specific or otherwise, duties of any kind, nature or description and undertakings of every kind or nature, contingent liabilities, bank/ corporate guarantees, duties, taxes, obligations under any licenses or permits or schemes and the all other liabilities of any description whatsoever, whether present or future, and howsoever raised or incurred



or utilized along with any charge, Encumbrance, lien or security thereon;

- xiii). All the employees of the Transferor Company; and
- xiv). All legal or other proceedings of whatsoever nature against the/ initiated by the Transferor Company.

All the terms, words, expressions which are used in this Scheme and not defined in this Scheme shall, unless repugnant or contrary to the context or meaning thereof, have the same meaning ascribed to them under the Act and other Applicable Law, rules, regulations, byelaws, as the case may be or any statutory modification or re-enactment thereof from time to time. In this Scheme, unless the context otherwise requires:

- the words denoting the singular shall include the plural and vice versa;
- headings and bold typefaces are only for convenience and shall be ignored for the purpose of interpretation;
- references to the word "include" or "including" shall be construed without limitation:
- iv. a reference to an article, clause, section, paragraph, annexure or schedule is, unless indicated to the contrary, a reference to an article, clause, section, paragraph, annexure or schedule of this Scheme;
- reference to a document includes an amendment or supplement to or replacement or novation of, that document; and
- vi. word(s) and expression(s) elsewhere defined in this Scheme will have the meaning(s) respectively ascribed to them.

2. DATE OF TAKING EFFECT AND OPERATIVE DATE

This Scheme as set out herein in its present form or with any modification(s), as may be approved or imposed or directed by the NCLT or made as per Clause 20 of this Scheme, as the case may be, shall have legal effect and force or be effective from the Appointed Date and shall be operative from the Effective Date.

3. SHARE CAPITAL OF THE COMPANIES

3.1. The authorised, issued, subscribed and paid-up share capital of Aleoas on February 28, 2022 is as under:











| Particulars | Amount (in Rs) |
|--|-------------------|
| Authorised share capital | |
| 5,00,000 Equity shares of Rs. 10 each | 50,00,000 |
| Total | 50,00,000 |
| Issued, subscribed and paid-up share capital | |
| 5,00,000 Equity shares of Rs. 10 each | 50,00,000 |
| Total | 50,00,000 |
| | - Ir |

There is no change in the authorised, issued, subscribed and paid up share capital of Aleor after the aforesaid date. Further, as on the date of approval of Scheme by Board of Directors of the Companies, the Transferee Company (along with its nominees) holds 100% of issued, subscribed and paid-up share capital of the Transferor Company.

There are 5,39,200 share warrants of Re. 1 each issued by the Transferor Company. The aforesaid share warrants are fully held by the Transferes Company.

3.2. The authorised, issued, subscribed and paid-up share capital of Alembic as on February 28, 2022 is as under:

| Particulars | Amount (in Rs) |
|--|-------------------|
| Authorised share capital | 1 |
| 20,00,00,000 Equity shares of Rs. 2 each | 40,00,00,000 |
| Total | 40,00,00,000 |
| Issued, subscribed and paid-up Share Capital | |
| 19,65,63,124 Equity shares of Rs. 2 each | 39,31,26,248 |
| Total | 39,31,26,248 |



There is no change in the authorised, issued, subscribed and paid-up share capital of Alembic after the aforesaid date.







PART III

4. TRANSFER AND VESTING OF THE UNDERTAKING OF TRANSFEROR COMPANY WITH THE TRANSFEREE COMPANY

- 4.1. Upon the Scheme being effective and subject to the provisions of this Scheme in relation to the mode of transfer and vesting, the Undertaking of the Transferor Company shall, without any further act, instrument or deed, be and stand transferred to and vested in, and/ or be deemed to have been and stand transferred to and vested in, the Transferee Company, on and from the Appointed Date, so as to become, the estate, assets, rights, title, interest and authorities of the Transferee Company as part of and consequent upon the amalgamation pursuant to provisions of Sections 230 to 232 of the Act and all other applicable provisions of the Act and in compliance with the provisions of Section 2(1B) of the Income-tax Act, 1961;
- 4.2. Without prejudice to the aforesaid, all the immovable property (including but not limited to the land, buildings, offices, factories, sites, tenancy rights related thereto, and other immovable property, including accretions and appurtenances), whether or not included in the books of the Transferor Company, whether freehold or leasehold (including but not limited to any other document of title, rights, interest and easements in relation thereto, and any shares in cooperative housing societies associated with such immovable property) shall stand transferred to and be vested in the Transferee Company, as successor to the Transferor Company, without any act or deed to be done or executed by the Transferor Company, as the case may be and/ or the Transferee Company;
- 4.3. Upon the Scheme being effective, with respect to the assets forming part of the Undertaking of the Transferor Company that are movable in nature or are otherwise capable of being transferred by manual delivery or endorsement and/ or delivery, the same shall stand transferred to and vested in, without any further act or execution of an instrument with the intent of vesting such assets in the Transferee Company as on the Appointed Date. The vesting pursuant to this sub-clause shall be deemed to have occurred by manual delivery or endorsement and/ or delivery, as appropriate to the property being vested, and the title to such property shall be deemed to have been transferred and vested accordingly. No additional stamp duty shall be payable on the transfer.







of such movable properties upon its transfer and vesting in the Transferee Company;

- 4.4. With respect to the assets of the Undertaking of the Transferor Company, other than those referred to in the Clauses above, including sundry debtors, bank balances, receivables, bills, credits, promissory notes, debit notes, outstanding loans and advances, if any, recoverable in cash or in kind or for value to be received, bank balances and deposits, if any, with government, semi-government, local and other authorities and bodies or with any company or other person etc., shall, without any further act, instrument or deed, with effect from the Appointed Date, be transferred to and vested in and/ or be deemed to be transferred to and vested in the Transferee Company pursuant to the provisions of Sections 230 to 232 and all other applicable provisions of the Act. It is hereby clarified that all the investments made by the Transferor Company shall, pursuant to Sections 230 to 232 and all other applicable provisions of the Act, if any, and the provisions of this Scheme, without any further act or deed, be transferred to and vested in or be deemed to have been transferred to and vested in the Transferee Company;
- 4.5. For the avoidance of doubt, it is clarified that upon the coming into effect of this Scheme, in accordance with the provisions of relevant laws. consents, permissions, licences, certificates, authorities (including for the operation of bank accounts), powers of attorney given by, issued to or executed in favour of the Transferor Company, and the rights and benefits under the same, and all quality certifications and approvals, trademarks, brands, patents and domain names, copyrights, industrial designs, trade secrets and other intellectual property and all other interests relating to the goods or services being dealt with by the Transferor Company shall be transferred to and vested in the Transferee Company without any further act or deed, and shall be appropriately mutated by the statutory authorities concerned therewith in favour of the Transferee Company. Upon the Scheme being effective, the benefit of all statutory and regulatory permissions, environmental approvals and consents, registrations or other licenses and consents shall vest in and become available to the Transferee Company, pursuant to the Scheme, on the same terms and conditions with effect from the Appointed Date. In so far as various incentives including export incentive, duty drawback, subsidies, rehabilitation schemes, special status, advance licenses and







65

other benefits or privileges enjoyed, granted by any Government body, local authority or by any person or availed of by the Transferor Company, are concerned, the same shall vest with and be available to the Transferee Company on the same terms and conditions;

- 4.6. Upon the Scheme being effective and subject to the other provisions of the Scheme, all contracts, deeds, bonds, agreements and other instruments of whatsoever nature to which the Transferor Company is a party subsisting or having effect on or immediately before the Effective Date shall remain in full force and effect against or in favour of the Transferee Company and shall be binding on and be enforceable by an i against the Transferee Company as fully and effectually as if the Transferee Company had at all times been a party thereto. Any inter-se contracts (including lease agreement) between the Transferor Company and the Transferee Company shall stand cancelled and cease to operate upon the coming into effect of this Scheme;
- 4.7. Upon the Scheme being effective, in so far as the various incentives, tax exemption, deductions, benefits, subsidies, grants, special status, registrations and other benefits or privileges enjoyed including in respect of income-tax [including Minimum Alternative Tax (MAT)], excise (including Modvat / Cenvat), customs, VAT, sales tax, service tax, goods and service tax etc., granted by any Appropriate Authority, or availed of by the Transferor Company are concerned, the same shall, without any further act or deed, vest with and be available to the Transferee Company on the same terms and conditions with effect from the Appointed Date
- 4.8. Any tax holiday/deduction/exemption/carry forward losses (including loss as per books of accounts) and unabsorbed depreciation (including unabsorbed depreciation as per books of accounts) entitled to/enjoyed/availed by the Transferor Company under the provisions of Income Tax Act, 1961 shall stand transferred to and vested in or deemed to be transferred to and vested in the Transferee Company and be entitled to /enjoyed/availed/utilised by the Transferee Company on and from the Appointed Date in the same manner as would have been entitled to /enjoyed/availed/utilised by the Transferee Company before implementation of this Scheme;





W.

نونا

- 4.9. Upon the Scheme being effective and with effect from the Appointed Date, all income, expense, debts, liabilities, including, without limitation, all secured and unsecured debts, sundry creditors. contingent liabilities, duties, obligations and undertakings of the Transferor Company, of every kind, nature and description whatsoever and howsoever arising, raised, incurred or utilised for its operations and activities, shall, pursuant to the sanction of this Scheme by the NCLT and under the provisions of Sections 230 to 232 and other applicable provisions, if any, of the Act, without any further act, instrument, deec. matter or thing being made, done or executed, be transferred to and vested in, or be deemed to have been transferred to and vested in, the Transferee Company and shall be assumed by the Transferee Company to the extent they are outstanding as on the Effective Date so as to become, as on and from the Appointed Date, the income, expenses, liabilities, debts, duties and obligations of the Transferee Company on the same terms and conditions as were applicable to the Transfercr Company, and the Transferee Company shall meet, discharge and satisfy the liabilities and it shall not be necessary to obtain the consent of any third party or other person who is a party to any contract or arrangement by virtue of which such liabilities have arisen in order to give effect to the provisions of this Clause;
- 4.10. All loans raised and utilised and all debts, liabilities, duties, and obligations incurred or undertaken by the Undertaking of the Transferor Company on or after the Appointed Date and prior to the Effective Date shall be deemed to have been raised, used, incurred or undertaken for and on behalf of the Transferee Company and to the extent they are outstanding on the Effective Date, shall, upon the coming into effect of this Scheme and under the provisions of Sections 230 to 232 of the Act, without any further act, instrument or deed be and stand transferred to and vested in the Transferee Company and shall become the loans and liabilities, duties and obligations of the Transferee Company which shall meet, discharge and satisfy the same;

4.11. Upon the Scheme being effective, where any of the liabilities and obligations of the Transferor Company subsisting as on the Appointed Date or which has arisen after the Appointed Date but prior to the Effective Date, have been discharged by the Transferor Company on the





after the Appointed Date and prior to the Effective Date, such discharge shall be deemed to have been for and on behalf of the Transferee Company;

- 4.12. The vesting of the Undertaking of the Transferor Company, as aforesaid, shall be subject to the Encumbrances, if any, over or in respect of any of the assets or any part thereof, provided however that such Encumbrances shall be confined only to the relevant assets of Transferor Company or part thereof on or over which they are subsisting on and vesting of such assets in Transferee Company and no such Encumbrances shall extend over or apply to any other asset(s) of Transferee Company. Any reference in any security documents or arrangements (to which Transferor Company is a party) related to any assets of Transferor Company shall be so construed to the end and intent that such security shall not extend, nor be deemed to extend, to any of the other asset(s) of the Transferee Company. Similarly, Transferee Company shall not be required to create any additional security over assets vested under this Scheme for any loans, debentures, deposits, or other financial assistance already availed of/ to be availed of by it, and the Encumbrances in respect of such indebtedness of Transferee Company shall not extend or be deemed to extend or apply to the assets so vested:
- 4.13. If and to the extent there are investments, loans, deposits, or balances inter-se between the Transferor Company and the Transferoe Company, the obligations in respect thereof shall, on and from the Appointed Date, come to an end and suitable effect shall be given in the books of the Transferee Company upon the Scheme being Effective;
- 4.14. Upon the Scheme being effective, any tax liabilities under the Incometax Act, 1961, Customs Act, 1962, Central Excise Act, 1944, value added tax laws, entertainment tax as applicable in any State in which the Transferor Company operates, Central Sales Tax Act, 1956, any other State Sales Tax laws/ Value Added Tax laws, or Service Tax laws, or Goods and Service Tax laws or Corporation Tax laws, or other applicable laws/ regulations dealing with taxes/ duties/ levies/cess (hereinafter in this Clause referred to as "Tax Laws") to the extent not provided for cr covered by tax provision in the Transferor Company's accounts made as on the date immediately preceding the Appointed Date shall be





D

transferred to the Transferee Company. Any surplus in the provision for taxation / duties/ levies account including advance tax, self assessment tax and tax deducted/collected at source as on the date immediately preceding the Appointed Date will also be transferred to the account of and belong to the Transferee Company:

- 4.15. Any amount including refund under the Tax Laws due to Transfercr Company consequent to the assessment proceedings or otherwise and which have not been received by the Transferor Company as on the date immediately preceding the Appointed Date shall also belong to and be receivable by the Transferee Company upon the Scheme being effective;
- 4.16. Without prejudice to the above, all benefits, including the benefits under Tax Laws, to which the Transferor Company is entitled to in terms of the applicable Tax Laws of the Union and State Governments, shall be available to and vest in the Transferee Company upon the Scheme being effective;
- 4.17. Pursuant to this Scheme becoming effective, the Transferee Company shall be entitled to secure the record of the change in the legal ownership upon the vesting of the Undertaking of the Transferor Company in accordance with the provisions of Sections 230 to 232 and all other applicable provisions, if any, of the Act. The Transferor Company and the Transferee Company shall be jointly and severally authorized to execute any writings and/ or carry out any formalities or compliances or do any act, thing or deed in this regard;
- 4.18. Without prejudice to the other provisions of this Scheme and notwithstanding the fact that vesting of the Undertaking of the Transferor Company occurs by virtue of this Scheme, the Transferoe Company may, at any time after the coming into effect of this Scheme, in accordance with the provisions hereof, if so required under any law or otherwise, take such actions and execute such deeds (including deeds of adherence), confirmations or other writings or tripartite arrangements with any party to any contract or arrangement to which the Transferor Company is a party, as may be necessary in order to give formal effect to the provisions of this Scheme. The Transferoe Company shall, under the provisions of this Scheme, be deemed to be authorized to execute any such writings on behalf of the Transferor Company to carry out or





perform all such formalities or compliances referred to above on the part of the Transferor Company, as the case may be, to be carried out cr performed; and

4.19. It is hereby clarified that if any assets (including but not limited to any estate, rights, title, interest in or authorities relating to such assets) in relation to the Transferor Company cannot be transferred to the Transferoe Company for any reason whatsoever, then directors of the Transferor Company shall hold such assets in trust for the benefit of the Transferee Company till such period the transfer is effectively carried out in favour of the Transferee Company.

5. STAFF, WORKMEN & EMPLOYEES

5.1. Upon this Scheme coming into effect and with effect from the Effective Date, the Transferee Company undertakes to engage all the employees of the Transferor Company on the terms and conditions not less favourable than those on which they are engaged by the Transfercr Company without any interruption of service as a result of the amalgamation of the Transferor Company with the Transferee Company. The Transferee Company also agrees that the services of all such employees with the Transferor Company prior to the amalgamation of the Transferor Company with the Transferee Company shall be taken into account for the purposes of all existing benefits to which the said employees may be eligible, including for the purpose of payment of any retrenchment compensation, gratuity and other retiral / terminal benefits and to this effect the accumulated balances, if any, standing to the credit of the employees in the existing provident fund, gratuity func. superannuation fund and such other fund of which they are members will be transferred to such provident fund, gratuity func, superannuation fund and such other fund nominated by the Transferee Company and/or such new provident fund, gratuity func, superannuation fund, such other fund to be established and caused to be recognized by the Appropriate Authorities, by the Transferee Company, or to the government provident fund in relation to the employees of the Transferor Company who are not eligible to become members of the provident fund maintained by the Transferee Company: and









FO

5.2. Pending the transfer as aforesaid, the provident fund, gratuity fund, superannuation fund and such other dues of the employees would be continued to be deposited in the existing provident fund, gratuity fund, superannuation fund and such other fund respectively of the Transferor Company.

LEGAL PROCEEDINGS

- 6.1. Upon the Scheme becoming effective, any suit, cause of actions, appeal or other legal, quasi-judicial, arbitral or other administrative proceedings of whatever nature ("Proceedings") by or against the Transferor Company pending and/ or arising before the Effective Date shall not abate or be discontinued or be in any way prejudicially affected by reason of the transfer of the Undertaking of the Transferor Company pursuant to this Scheme or by anything contained in this Scheme but the Proceedings shall be continued, prosecuted and enforced by or against the Transferee Company, as the case may be in the same manner and to the same extent as they would or might have been continued, prosecuted and enforced by or against the Transferor Company as if the Scheme had not been made. On and from the Effective Date, the Transferee Company shall and may continue any Proceedings which were earlier in the name of the Transferor Company;
- 6.2. The Transferee Company undertakes to have, all respective legal or other Proceedings initiated by or against the Transferor Company, transferred into its name and to have the same continued, prosecuted and enforced by or against the Transferee Company, as the case may be, to the exclusion of the Transferor Company; and
- 6.3. The transfer and vesting of the Undertaking of the Transferor Company under the Scheme and the continuation of the Proceedings by or against the Transferee Company under Clause 6.1 and 6.2 above, shall not affect any transaction or Proceedings already completed by the Transferor Company on or after the Appointed Date and prior to this Scheme becoming effective to the end and intent that the Transferor Company accepts and adopts all acts, deeds and things done and executed by the Transferor Company as acts, deeds and things being done and executed by and on behalf of the Transferee Company.







7. VALIDITY OF EXISTING RESOLUTIONS, ETC

Upon the coming into effect of this Scheme, the resolutions of the Transferor Company, as are considered necessary by the Board of Directors of the Transferee Company and which are valid and subsisting on the Effective Date shall continue to be valid and subsisting and be considered as resolutions of the Transferee Company and if any such resolutions have any monetary limits approved under the provisions of the Act, or any other applicable statutory provisions, then said limits, as are considered necessary by the Board of Directors of the Transferee Company, shall be added to the limits, if any, under like resolutions passed by the Transferee Company and shall constitute the aggregate of the said limits in the Transferee Company.

CONTRACTS, DEEDS, ETC.

- 8.1. Notwithstanding anything to the contrary contained in the contract, deed, bond, agreement or any other instrument, but subject to the other provisions of this Scheme, all contracts, deeds, bonds, agreements and other instruments, if any, of whatsoever nature and subsisting or having effect on the Effective Date and relating to the Transferor Company shall continue in full force and effect against or in favour of the Transferee Company and may be enforced effectively by or against the Transferee Company as fully and effectually as if, instead of the Transferor Company, the Transferee Company had been a party thereto; and
- 8.2. The Transferee Company shall at any time after the coming into effect of this Scheme in accordance with the provisions hereof, if so required, under any law or otherwise, enter into, or issue or execute deeds, writings, confirmations, novations, declarations, or other documents with, or in favour of any party to any contract or arrangement to which the Transferor Company is a party or any writings as may be necessary to be executed in order to give formal effect to the above provisions. Transferee Company shall, be deemed to be authorized to execute any such writings on behalf of Transferor Company and to carry out or perform all such formalities or compliances required for the purposes referred to above on the part of Transferor Company.





9. PERMITS

With effect from the Appointed Date, all the permits held or availed cf by, and all rights and benefits that have accrued to, the Transferor



Company, pursuant to the provisions of Sections 230 to 232 of the Act, shall without any further act, instrument or deed, be transferred to, and vested in, or be deemed to have been transferred to, and vested in, and be available to, the Transferce Company so as to become, from the Effective Date, the permits, estates, assets, rights, title, interests and authorities of the Transferce Company and shall remain valid, effective and enforceable on the same terms and conditions to the extent permissible in Applicable Laws. Upon the Appointed Date and until the permits are transferred, vested, recorded, effected, and/or perfected, in the record of the Appropriate Authority, in favour of the Transferce Company, the Transferce Company is authorized to carry on business in the name and style of the Transferor Company and under the relevant license and/or permit and/or approval, as the case may be, and the Transferce Company shall keep a record and/or account of such transactions.

10. TAX CREDITS

- 10.1. The Transferee Company will be successor of the Transferor Company. Hence, it will be deemed that the benefit of any tax credits whether central, state or local, availed by the Transferor Company and the obligations if any for payment of tax on any assets of the Transferor Company shall be deemed to have been availed by the Transferor Company or as the case may be deemed to be the obligations of Transferee Company. As the Scheme does not contemplate removal of any asset by the Transferee Company from the premises in which it is installed, no reversal of any tax credit needs to be made;
- 10.2. With effect from the Appointed Date and upon the Scheme becoming effective, all taxes, duties, cess payable/ receivable by the Transferor Company including all or any refunds/ credit including MAT/ claims relating thereto shall be treated as the asset/ liability or refunds/ credit, including MAT/ claims as the case may be, of the Transferee Company. The Transferee Company shall be entitled to get credit/ claim refund of any tax paid and/ or tax deduction at source or tax collected at source on or after the Appointed Date by the Transferor Company; and
- 10.3. In order to give effect of this Scheme, the Transferor Company and the Transferee Company shall be entitled to, amongst others, file/revise its income tax returns, TDS/ TCS returns, wealth tax returns, service tax.





excise duty, sales tax, value added tax, goods and services tax, entry tax, cess, professional tax or any other statutory returns, if required, claim credit for advance tax paid, tax deducted at source, tax collected at source, claim for sum prescribed under Section 43B of the Income-tex Act, 1961 on payment basis, claim for deduction of provisions written back by the Transferee Company previously disallowed in the hands of Transferor Company under the Income-tax Act, 1961, credit of tax under Section 115JB read with Section 115JAA of the Income-tax Act, 1961. credit of foreign taxes paid/ withheld etc., if any, as may be required consequent to implementation of this Scheme, even if the prescribed time limits for filing or revising such returns have lapsed without incurring any liability on account of interest, penalty, late fee or any other sum. Transferee Company shall have the right to claim refunds. tax credits, set-offs and/ or adjustments relating to its income or transactions entered into by it, by virtue of this Scheme with effect from Appointed Date. The taxes or duties paid by, for, or on behalf of the Transferor Company relating to the period on or after Appointed Date shall be deemed to be the taxes or duties paid by the Transferee Company, and accordingly the Transferee Company shall be entitled to claim credit or refund for such taxes or duties.

11. SAVING OF CONCLUDED TRANSACTIONS

Subject to the terms of this Scheme, the transfer of the Undertaking. licences, permits, registrations, memberships and approvals as specified under this Scheme and the continuance of Proceedings by or against the Transferee Company under Clause 6 above shall not affect any transaction or Proceedings already concluded by the Transferor Company on or before the Appointed Date, or after the Appointed Date till the Effective Date, to the end and intent that the Transferee Company accepts and adopts all acts, deeds and things made, done and executed by the Transferor Company as acts, deeds and things made, done and executed by or on behalf of the Transferee Company.

12. CANCELLATION OF SHARE CAPITAL, SHARE WARRANTS AND NOT-CONVERTIBLE DEBENTURES OF TRANSFEROR COMPANY

The Transferee Company (along with its nominees) holds 100% of the issued, subscribed and paid-up share capital of the Transferor Company. Accordingly, the Transferor Company is the wholly owned subsidiary of the Transferee Company. Upon the Scheme becoming





Q.



effective, no shares of the Transferee Company shall be allotted in lieu or exchange of the holding of the Transferee Company in the Transferer Company and the share capital of the Transferor Company shall stand cancelled on the Effective Date without any further act, instrument, cr deed.

Further, upon the Scheme becoming effective, share warrants and Non-Convertible Debentures issued by the Transferor Company to the Transferee Company shall stand cancelled on the Effective Date without any further act, instrument, or deed.

13. ACCOUNTING TREATMENT IN THE BOOKS OF TRANSFEREE COMPANY

Notwithstanding anything to the contrary herein, upon this Scheme becoming effective, the Transferee Company shall give effect to the accounting treatment in the books of accounts in accordance with the Accounting Standards specified under Section 133 of the Act read with the Companies (Indian Accounting Standards) Rules, 2015, or any other relevant or related requirement under the Act, as applicable on the Appointed Date.

Accordingly, the Transferor Company and Transferee Company, being entities under common control, the accounting would be done at Transferor Company carrying amounts as on the Appointed Date for all the assets and liabilities acquired by the Transferee Company of the Transferor Company by applying the principles as set out in Appendix C of Ind AS 103 'Business Combinations' and inter-company balances and inter-company investments, if any, between Transferor Company and Transferee Company shall stand cancelled.

14. DISSOLUTION OF THE TRANSFEROR COMPANY

On and from the Effective Date, the Transferor Company shall stand dissolved without being wound up.

15. OPERATIONS, ACTIVITIES AND PROPERTIES IN TRUST AND CONDUCT OF OPERATIONS FOR TRANSFEREE COMPANY

Unless otherwise stated herein under, with effect from the Appointed Date and up to and including the Effective Date:











- 15.1. The Transferor Company shall be deemed to have been carrying on and shall carry on its operations and activities and shall be deemed to have held and stood possessed of and shall hold and stand possessed of the Undertaking of the Transferor Company for and on account of, and in trust for the Transferee Company and shall account for the same to the Transferee Company. The Transferor Company hereby undertakes to hold the said Undertaking with utmost prudence until the Effective Date;
- 15.2. The Transferor Company shall carry on the business and activities, with reasonable diligence and business prudence;
- 15.3. All the surplus, income, taxes (including MAT credit, tax paid and tax deducted/ collected at source) or any costs, charges, expenditure, accumulated losses accruing to the Transferor Company or expenditure or losses arising or incurred or suffered by the Transferor Company shall for all purpose be treated and be deemed to be and accrue as the surplus, income, taxes (including MAT credit, tax paid and tax deducted/ collected at source) or any costs, charges, expenditure or accumulated losses of the Transferee Company, as the case may be;
- 15.4. The Transferor Company shall not vary the terms and conditions of employment of any of their employees, without the prior consent in writing of any of the persons authorized by the Board of Directors of the Transferee Company, except in the ordinary course of business; and



15.5. The Transferee Company shall be entitled, pending sanction of the Scheme, to apply to the concerned authorities and all other agencies, departments and authorities concerned as are necessary under any law for such consents, approvals, and sanctions which the Transferee Company may require to carry on the operations of the Transferer Company and to give effect to the Scheme.



16. APPLICATIONS TO NCLT OR OTHER APPROPRIATE AUTHORITIES

16.1. The Transferor Company and the Transferee Company shall, with all reasonable dispatch, make necessary application under Sections 230 to 232 and/ or other applicable provisions of the Act to the NCLT or such other Appropriate Authority, for seeking order for dispensing with cr convening, holding and conducting of, meeting of the shareholders and/or creditors of the Transferor Company and the Transferee

Company, as may be directed by NCLT or such other Appropriate Authority for sanction of this Scheme and all matters ancillary or incidental thereto: and

16.2. On the Scheme being approved by the requisite majorities of the shareholders and/ or creditors of the Transferor Company and the Transferee Company, whether at meeting or otherwise, as prescribed under the law and/ or as directed by the NCLT or such other Appropriate Authority, the Transferor Company and the Transferee Company shall, with all reasonable dispatch, apply to the NCLT or such other Appropriate Authority for sanctioning of the Scheme under Sections 230 to 232 of the Act, and for such other order or orders, as the NCLT or such other Appropriate Authority may deem fit for carrying this Scheme into effect.

17. CONSOLIDATION OF AUTHORISED SHARE CAPITAL

17.1. Upon this Scheme becoming effective, the authorised share capital of the Transferor Company amounting to INR 50,00,000 (Rupees Fifty Laka Only), as mentioned in Clause 3.1 above, shall be added to the authorised share capital of the Transferee Company, as on the Effective Date, without any further act or deed and without any further payment of any stamp duty or registration fees. For this purpose, the stamp duty and fees already paid on the authorised share capital of the Transferor Company shall be utilised and applied to the increased authorised share capital of the Transferee Company and there would be no requirement for any other further payment of stamp duty and/ or fee by the Transferee Company for increase in the authorised share capital to that extent. The authorised share capital of the Transferee Company will thus be increased to that effect by virtue of the Scheme becoming effective and no separate procedure shall be required to be followed under the Act;



17.2. Consequent upon the amalgamation, the authorised share capital of the Transferee Company will be amended / altered / modified as under:

| Authorised Share Capital | Amount (in Rs. | | |
|--|----------------|--|--|
| 20,25,00,000 Equity shares of Rs. 2 each | 40,50,00,000 | | |
| Total | 40,50,00,000 | | |







FF

11.

17.3. Consequently, Clause V of the Memorandum of Association of the Transferee Company shall, without any act, instrument, or deed, be and stand altered, modified, and amended pursuant to applicable provisions of the Act. The revised Capital Clause of the Transferee Company shall be as under:

"The Authorised Share Capital of the Company is Rs. 40,50,00,000 (Rupees Forty Crores Fifty Lakhs only) divided into 20,25,00,000 (Twenty Crores and Twenty Five Lakhs) Equity Shares of Rs. 2 (Rupees Two) each."

17.4. It is clarified that the approval of the shareholders of the Transferee Company to the Scheme shall be deemed to be their consent/approval also to the consequential alteration of the Memorandum of Association of the Transferee Company under Sections 13, 61 and 64 of the Act and other Applicable Laws and the Transferee Company shall not be required to seek separate consent/approval of its shareholders for such alteration of the Memorandum of Association of the Transferee Company.







PART IV

18 SCHEME CONDITIONAL ON APPROVALS / SANCTIONS

- 18.1 The Scheme is and shall be conditional upon and subject to the followings:
 - a) The Scheme being approved by requisite majority in numbers and value of such classes of persons including the shareholders and/or creditors of the Transferor Company and the Transferee Company, as required under the Act or as may be directed by the NCLT and/or any other competent authority as may be applicable;
 - b) The requisite consent, approval or permission of any other statutory or regulatory authority, which by law may be necessary for the implementation of this Scheme;
 - c) The Scheme being sanctioned by the NCLT under Sections 230 to 232 and all other applicable provisions of the Act and/ or any Appropriate Authority; and
 - d) Certified Copies of the Order of the NCLT or such other competent authority, as may be applicable, sanctioning this Scheme being filed with the Registrar of Companies having jurisdiction over the Companies.

19 EFFECT OF NON-RECEIPT OF PERMITS AND MATTERS RELATING TO REVOCATION/ WITHDRAWAL OF THIS SCHEME

- 19.1 The Transferor Company and the Transferee Company acting through their respective Boards shall each be at liberty to withdraw from this Scheme: (a) in case any condition or alteration imposed by any Appropriate Authority is unacceptable to any of them; or (b) they are of the view that coming into effect of this Scheme could have adverse implications on the respective Companies;
- 19.2 In the event of revocation/ withdrawal under Clause 19.1, no rights and liabilities whatsoever shall accrue to or be incurred inter se the Transferor Company and the Transferee Company or their respective shareholders or creditors or employees or any other person, save and except in respect of any act or deed done prior thereto as is contemplated hereunder or as to any right, liability or obligation which has arisen or accrued pursuant thereto and which shall be governed and be preserved or worked out as is specifically provided in the Scheme or in accordance







with the Applicable Law and in such case, each Company shall bear its own costs, unless otherwise mutually agreed; and

19.3 If any Clause of this Scheme is held to be invalid, ruled illegal by any Court of competent jurisdiction, or unenforceable under present or future laws, then it is the intention of the Companies that such Clause shall be severable from the remainder of this Scheme, and this Scheme shall not be affected thereby, unless the deletion of such Clause shall cause this Scheme to become materially adverse to any party, in which case the Board of Directors of the Companies involved in the Scheme shall attempt to bring about a modification in this Scheme, as will best preserve for the parties the benefits, and obligations of this Scheme, including, but not limited to, such Clause.

20 MODIFICATIONS/ AMENDMENTS TO THE SCHEME

- 20.1 The Boards of the Transferor Company and Transferee Company acting themselves or through authorized persons, may consent jointly but not individually, on behalf of all persons concerned, to any modifications or amendments of this Scheme at any time and for any reason whatsoever, or to any conditions or limitations that the NCLT or any other Appropriate Authority may deem fit to direct or impose or which may otherwise be considered necessary, desirable or appropriate by the Board of the Transferor Company and the Transferee Company and solve all difficulties that may arise for carrying out this Scheme and do all acts, deeds and things necessary for putting this Scheme into effect; and
- 20.2 For the purpose of giving effect to this Scheme or to any modification thereof, the Boards of the Transferor Company and the Transferoe Company acting themselves or through authorized persons may jointly but not individually, give and are jointly authorized to give such directions including directions for settling any question of doubt or difficulty that may arise and such determination or directions, as the case may be, shall be binding on all parties, in the same manner as if the same were specifically incorporated in this Scheme. It is clarified that individual companies acting themselves or through authorized persons may individually approach the NCLT or any other Appropriate Authority to seek clarifications for implementation of the Scheme.



W.



21 GENERAL TERMS AND CLAUSES

- 21.1 Upon the Scheme becoming effective, the Transferor Company and Transferee Company shall have the right to revise their respective financial statements and returns along with prescribed forms, filings. and annexures under the Tax Laws and/ or Applicable Laws to give effect to the provisions of the Scheme. In order to give effect of this Scheme, the Transferor Company and the Transferee Company shall be entitled to, amongst others, file and/ or revise its income tax returns, TDS/ TCS returns, wealth tax returns, service tax, excise duty, sales tax, value added tax, goods and services tax, entry tax, cess, professional tax or any other statutory returns, if required, claim credit for tax paid, tax deducted at source, tax collected at source, claim for sum prescribed under Section 43B of the Income-tax Act, 1961 on payment basis, claim for deduction of provisions written back by Transferee Company previously disallowed in the hands of Transferor Company under the Income-tax Act, 1961, credit of tax under Section 115JB read with Section 115JAA of the Income-tax Act, 1961, credit of foreign taxes paid. withheld etc. if any, as may be required consequent to implementation of this Scheme, even if the prescribed time limits for filing or revising such returns have lapsed without incurring any liability on account of interest, penalty, late fee or any other sum. Such returns shall be filed based on reconstructed accounts drawn up with effect from the Appointed Date and any tax including MAT shall be computed accordingly. Transferee Company shall have the right to claim refunds, tax credits, setoffs and/ or adjustments relating to its income or transactions entered into by it, by virtue of this Scheme with effect from Appointed Date. The taxes or duties paid by, for, or on behalf of the Transferor Company relating to the period on or after Appointed Date shall be deemed to be the taxes or duties paid by the Transferee Company, and accordingly the Transferee Company shall be entitled to claim credit or refund for such taxes or duties;
- 21.2 Upon this Scheme becoming effective, the Transferee Company is expressly permitted to prepare and/or revise their financial statements and returns along with prescribed forms, filings and annexures under the Income-tax Act, 1961 (including for minimum alternate tax purposes and tax benefits), service tax law, GST law and other tax laws, and to claim refunds and/or credits for taxes paid (including minimum alternate tax), and to claim tax benefits (including brought forward book.







losses) under the Income-tax Act, 1961, etc., and for matters incidental thereto, if required to give effect to the provisions of this Scheme. The order of the jurisdictional NCLT sanctioning the Scheme shall be deemed to be an order of the Competent Authority permitting the Transferee Company to prepare and/or revise its financial statements and books of accounts and no further act shall be required to be undertaken by the Transferee Company;

- 21.3 It is hereby clarified that submission of the Scheme to the NCLT and to Appropriate Authorities for their respective approvals is without prejudice to all rights, interests, titles, or defences that the Transferor Company and the Transferee Company may have under or pursuant to all Applicable Law; and
- 21.4 On the approval of this Scheme by the shareholders of the Transferor Company and the Transferee Company and such other classes of persons of the said Companies, if any, such shareholders, and classes of persons shall also be deemed to have resolved and accorded all relevant consents under the Act and Applicable Law or otherwise to the same extent applicable in relation to the amalgamation, related matters and this Scheme itself.

22 COSTS, CHARGES AND EXPENSES

All costs, charges and expenses including stamp duty and registration fee of any deed, document, instrument or NCLT's order including this Scheme or in relation to or in connection with negotiations leading up to the Scheme and of carrying out and implementing the terms and provisions of this Scheme and incidental to the completion of arrangement in pursuance of this Scheme shall be borne and paid by the Transferee Company.







Dermatology Products for Better Health



List of Assets of Aleor Dermaceuticals Limited, the Transferor Company, as on 31st July 2022 to be transferred to Alembic Pharmaceuticals Limited, the Transferee Company, pursuant to the scheme sanctioned by the Hon'ble National Company Law Tribunal, Bench at Ahmedabad.

Schedule

Part I

Particulars of Freehold Properties

(i) Land: Nil

(ii) Building:

All that industrial constructed areas and building admeasuring approx. 34,538 sq. mt. built up area located at City Survey No. 790/p, 779/p, Village: Karkhadi, Taluka: Padra, District: Vadodara - 331 440, Gujarat bounded as follows:

On or towards the north: 18 mt. canal road

On or towards the south: City Survey No. 733/p, 800, 785, Alembic F-3

On or towards the east: 18 mt. wide Naliya Road

On or towards the west: City Survey No. 818/b, 790/p, Alembic F-3

(iii) Plant and Machinery: (if any, imbedded or attached to earth): As per Annexure A.







Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadodara 390 003, Gujarat, India.
CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail : aleor@aleorderm.com | www.aleorderm.com







Part II

Particulars of Leasehold Properties

(i) Land: Nil

(ii) Building:

- (a) All that part and parcel of the 5 (five) storied industrial premises with built-up area of 15,362 sq. ft. (including common area and balconies), out of total built up area of 17,700 sq. ft. built on Plot No. 45 admeasuring 648 sq. yd. or equivalent to 541.72 sq. mtr. in Survey No.: 342 of ALEAP Industrial Estate situated in Gajularamaram village, Qutbullapur Mandal, Medchal - Malkajgiri district - 500 090, State of Telangana with the floor wise area detailed as below:
 - Ground Floor: 612 sq. ft. (Out of 2,950 sq. ft.)
 - 2. First Floor: 2,950 sq. ft.
 - 3. Second Floor: 2,950 sq. ft.
 - 4. Third Floor: 2,950 sq, ft.
 - 5. Fourth Floor: 2,950 sq. ft.
 - 6. Fifth Floor: 2,950 sq. ft.

The same is bounded by:

On or towards the north: Plot No. 44

On or towards the south: 18 mtr. wide Road No. 4 On or towards the east: 12 mtr. wide Road No. 5

On or towards the west: Plot No. 46

(iii) Plant and Machinery: (if any, imbedded or attached to earth): Nil







Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadodara 390 003, Gujarat, India.
CIN: U24302GJ2016FLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail : alear@alearderm.com | www.alearderm.com



Part III

A. Particulars of Investment in Shares & Securities:

| No. of shares | Description | Face Value (Amount in Rs.) per share | Current Market Value (Amount in Rs.) |
|------------------|---|--------------------------------------|--|
| 2,25,000 | 1% Redeemable Cumulative Non-Convertible Preference Shares of Enviro Infrastructure Company Limited | 10/- | 22,50,000/- |

B. Particulars of Bank Accounts:

| Sr. No. | Bank & Branch | Type of Account | Account No. |
|------------|--|----------------------|------------------|
| 1 | HDFC Bank Ltd. Alkapuri, Vadodara | Cash Credit | 00330310001159 |
| 2 | HDFC Bank Ltd. Lakdikapul, Hyderabad | Cash Credit | 00210310006004 |
| 3 | Yes Bank Ltd. Race Course Circle, Vadodara | Cash Credit | 000981400000183 |
| 4 | Yes Bank Ltd. Race Course Circle, Vadodara | Private Placement | 000981400000 |
| 5 | IDBI Bank Ltd. Race Course Circle, Vadodara | Current Account | 0378103000002707 |



Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadodara 390 003, Gujarat, India.

CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: alear@aleorderm.com | www.aleorderm.com



MLEOR

C. Registration with Various Authorities under respective laws, Bodies etc.:

As per Annexure B.

D. VEHICLES:

Vehicle Description - Bolero Ambulance ABS ABG Registration number - GJ06BT0872

For Aleor Dermaceuticals Limited

Mitanshu Shah Director

Encl.: Annexure A & B











Annexure A

Plant and Machinery: (imbedded or attached to earth):

 Located at the manufacturing facilities of Aleor Dermaceuticals Limited at City Survey No. 790/p, 779/p, Village: Karakhadi, Taluka: Padra, District: Vadodara - 331 440, Gujarat:

| Asset Description | |
|--|-----|
| 1000 kgs cream plant (nflp) (cortico) | *** |
| 1000 kgs cream plant (nflp) (general) | |
| 1000 kgs storage vessels (cortico) - 8 nos | |
| 13 door interlock system - 2 nos | |
| 15 kl diesel tank - 2 nos | |
| 2 door interlock system - 17 nos | |
| 2 door interlock system - 8 nos | |
| 20 kva online ups 42 ah - 3 nos | |
| 20 kva online ups 65 ah | |
| 20 kva online ups 65 ah - 2 nos | |
| 20 kva online ups 65 ah - 4 nos | |
| 20000 ltr tank - 2 nos | |
| 250 kgs cream plant (flp) (cortico) | |
| 250 kgs cream plant (flp) (general) | |
| 250 kgs cream plant (nflp) (immuno) | |
| 250 w power amplifier - 4 nos | |
| 3 door interlock system - 2 nos | |
| 3 door interlock system - 5 nos | |
| 30 kva online ups 75 ah | |
| 4 door interlock system - 8 nqs | |
| 4 in 1 online analyser | |
| 50 kgs cream plantpilot (flp) (general) | |
| 561kwp solar roof top | |
| 6 kva online ups | |
| 7 door interlock system - 10 nos | |
| 8 door interlock system - 4 nos | |
| Active repeater panel for fire | |
| Admin lift | |
| Advance oxidation system | |
| Ahu - 71 nos | |
| | LAC |



Aleor Dermaceuticals Ltd

Air blowers for aeration tank-1 - 3 nos

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Rood, Vadodara 390 003, Gujorat, India.

CIN: U24302G/2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: alear@alearderm.com | www.alearderm.com







| Asse | t | Description |
|---------|---|-------------|
| tanle 2 | 2 | 200 |

Air blowers for aeration tank-2 - 3 nos Air blowers for equalisation tank - 3 nos

Air compressor - 2 nos

Annunciator panel - 2 nos

Ao dd pump of wilden pump

Apfc panel - 2 nos

Auto transformer, vpi 3ph 100kva

Auto tube cleaning system

Autoclave 1801 praano

Automated screw screen

Automatic sliding gate

Automatic tube filling machine type pg-t

Axile fan 10000cfm

Axile fan 3000cfm - 2 nos

Battery analyzers - fluke

Belt oil skimmer (ogt) - 1

Belt oil skimmer (ogt) - 2

Bms system

Capsule filtration system for solvents

Ceiling suspended ahu

Ceiling suspended unit - ra 1549 cfm

Chemical (pe) dosing tank with agitator

Chemical (pe) dosing tank with agitator - 2 nos

Chemical dosing (lime) pumps - 2 nos

Chemical dosing (pe) pumps - 2 nos

Chemical dosing (pe) pumps - 4 nos

Chemical dosing pumps - 2 nos

Chemical dosing tank with agitator

Chemical dosing tank with agitator - 2 nos

Chilled water casset unit 1 tr - 17 nos

Chilled water casset unit 1 tr - 2 nos

Chilled water casset unit 1.3 tr - 2 nos

Chilled water casset unit 1.6 tr - 2 nos

Chilled water casset unit 2 tr - 42 nos

Chilled water casset unit 2.5 tr - 8 nos

Chilled water highwall unit 1.3 tr

Chilled water highwall unit 1.3 tr - 26 nos

Chilled water pumping skid

Chiller of 380 tr





Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadodara 390 003, Gujarat, India.

CIN: U24302GJ2016PLC09213? | PAN No: AACCA44791 | Tel.: 0265 2280550 | E-mail: aleor@aleorderm.com | www.aleorderm.com





| Ass | et Description |
|------------------------------------|--|
| Chiller of 380 tr - 3 nos | |
| Chimeny for hwg - 3 nos | |
| Cip for storage vessels (cortico) | |
| Cip for storage vessels (general) | |
| Cip system 70 l (for lobe pump | cleaning) |
| Circulation pump for hwg rsw 1 | 90 - 4 nos |
| Circulation pump for hwg rsw 3 | 50 - 2 nos |
| Co2 follding system | |
| Cod digestor | |
| Collapsible gate with track motor | or |
| Compressed air booster | |
| Compressed air receiver tank | |
| Compressor cooling water pump | skid |
| Constant temp water bath - 2 n | os |
| Control panel for hwg rsw 190 - | - 14-54-3-704-1 |
| Control panel for hwg rsw 350 - | 4 nos |
| Control panel for make up wate | |
| Control panel for pre treatment | All and the second seco |
| Control panel for storage tank a | |
| Control system-critical equip me | |
| Cooling tower of 255m3/hr - 3 r | nos |
| Cooling tower pumping skid | |
| Decanter feed pumps - 2 nos | |
| Dell server | |
| Dg set 1500 kva - 2 nos | |
| Digital water flow meter | |
| Diseal fire pump | |
| Diseal pump skid | |
| Dispax-reactor | |
| Dosing tank station with dda 30 | 1-4 ar |
| Drain pumps - 2 nos | |
| Dust extractor (cortico) | |
| Dust extractor (cortico) - 2 nos | |
| Dust extractor (general) - 2 nos | |
| Dust extractor (immuno) | |
| Dust extractor and table for filir | - Contract of the Contract of |
| Dynamic passbox - 2 nos - aero | sol |
| Dynamic passbox - 3 nos | 1 |



Aleor Dermaceuticals Ltd

Dynamic passbox - cfil

Registered Office : 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadadara 390 003, Gujarat, India.

CIN: UZ4302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail : aleor@alearderm.com | www.aleorderm.com



| Asset Description | |
|---|---|
| Dynamic passbox - floor mounted - aerosol | |
| Dynamic passbox - gfil | |
| Dynamic passbox with uv lights - cfil | |
| Dynamic passbox with uv lights - gfil | |
| Eau - 33 nos | |
| Effluent disposal pumps - 2 nos | |
| Effluent transfer pumps - 2 nos | |
| Effluent transfer pumps - 4 nos | |
| Electric installation - first floor new stabiltiy | |
| Electrical insta - ff doc area qa/qc/pro | |
| Electrical installation & equipments | - |
| Electrical installation & equipment - aerosol | |
| Electrical panel - etp | |
| Electrical panel - security block | |
| Electronic overhead homoginizer | |
| Electronic overhead stirrer | |
| Encloser top mount exhaust style | |
| Etp feed pump - 2 nos | |
| Etp inlet pumps - 2 nos | |
| Etp permit tank reject ro 120m3/day | |
| Faast system | |
| Feed water tank for hwg - 2 nos | |
| Filter cleaning booth - 3 nos | , |
| Filter media for sludge drying bed - 4 nos | |
| Fire aspiration system | |
| Fire hydrant electrical panel | |
| 774 March 174 William (1995) (1995) (1995) (1995) | |





Fire hydrant line Fire panel Fix type oxygen sensor with display - 5 nos Fixed type lel sensor - 3 nos

Flame proof hooter 80db - 6 nos Flap barrier - 3 nos

Flash mixer agitator

Flocculator agitator

Flow indicator

Flow indicator - 2 nos

Flow indicator - 4 nos

Flow indicator & totaliser

Flow meter for hsd - 2 nos

Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadodara 390 003, Gujarat, India.

CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: aleor@aleorderm.com | www.aleor.





| Asset Description | |
|---|-----|
| Flow totalisers (4-20 ma) | |
| Flow totalisers (4-20 ma) - 2 nos | |
| Flow totalisers (4-20 ma) - 6 nos | |
| Free chlorine analyser | |
| Gas piping manifold installation - aerosol | |
| General plant & machinery | |
| General plant & machinery - aerosol | |
| General plant & machinery - cfil | |
| General plant & machinery - cmfg | _ |
| General plant & machinery - eng | |
| General plant & machinery - etp | |
| General plant & machinery - gfil | 100 |
| General plant & machinery - gmfg | |
| General plant & machinery - ifil | |
| General plant & machinery - imfg | |
| Goods cum passenger elevator - 1 | |
| Goods cum passenger elevator - 2 | |
| Hdpe tank - 0.5kl | |
| Hdpe tank - 1kl | |
| Hdpe tank - 5kl - 2 nos | |
| Hot water generator rsw 190 - 4 nos | |
| Hot water generator rsw 350 - 2 nos | |
| Hsd pump | |
| Hsd storage tank | |
| Hvac - first floor new stabiltiy chmaber area | |
| Hvac system | |
| Hvac system - aerosol | |
| Hybrid ro membrane sys. 150m3/day | |
| Hydraulic drive scissor lift | |
| Indoor type 11kv ht vcb panel | |
| Jockey fire pump | |
| Lime prepartion tank agitator | |
| Main fire pump | |
| Main lighting panel | |
| Main lt panel | |
| Makeup water pump for hwg | |
| Mass flow meter | |
| Mass flowmeter | |



Aleor Dermaceuticals Ltd

Mcc panel - 9 nos

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadodara 390 003, Gujarat, India. CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: alear@alearderm.com | www.



| Asset Description | |
|---|---|
| Modular datacenter infrastructure smart | 2 |
| Ms tank - 2kl - 4 nos | |
| Nitrogen generator | |
| Online gas detector system - 4 nos | |
| Openscape business v2 x8 system | |
| Outdoor type 11kv ht vcb panel | |
| Oxygen gas detector | |
| P&f goods cum passanger lift - 1000 kg | |
| Pallet stretch wrapping machine | |
| Pcc panel - 2 nos | |
| Pid controller - 2 nos | |
| Plc panel for etp ro | |
| Plc panel for process cooling | |
| Potable heated water pumping skid | |
| Poweredge r440 server | |
| Pre treatment plant | |
| Premium vortex mixer - 4 nos | |
| Pressure sand filter - frp | |
| Primary sludge pumps | |
| Process cooling pumping skid-cort | |
| Process cooling pumping skid-gen | |
| Process effluent transfer pump - 2 nos | |
| Purified water generation system | |
| Remote for pws io panel | |
| Ro water makeup tank 1000 ltr | |
| Safety shower | |
| Scissor lift- 2000kg | |
| Secondar sludge pumps - 4 nos | |
| Secondary hot water pumping skid | |
| Self-propelled order picker lift | |
| Server & network system | |
| Sludge dewatering unit decanter | |
| Solar plant 250 kwp | |
| Ss tank - 1kl | |
| Ss tank - 5kl | |
| Static passbox | |
| | |





Aleor Dermaceuticals Ltd

Storage cabinet, 20-25°c, 1000 1 - 2 nos

Static passbox - (general)

Storage emc vne 3200

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadodara 390 003, Gujarat, India.

CIN: U24302GJ2016FLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: alear@alearderm.com | www.alearderm.com



| Asset Description | |
|--|--|
| Storage tank for pws | |
| Stp plant | |
| Sub distribution panel | |
| Tanker unloading system | |
| Temp monitoring control panel - 10 nos | |
| Transformer 2000 kva - 2 nos | |
| Treated effluent disposal pumps - 2 nos | |
| Vfd 15hp | |
| Vfd 7.5hp | |
| Vfd for chiller | |
| Vfd panel for chiller - 2 nos | |
| Vfm | |
| Walk in cooling cabinet,2-8°c,5000 l | |
| Walk in stability chamber - 10000 l - 3 nos | |
| Walk in stability chamber - 50000 l - 5 nos | |
| Walk in stability chamber - 5400 l - 4 nos | |
| Water chiller capacity 2.6 kw | |
| Water cooled chiller panel | |
| Wtp ro reject transfer pum - 2 nos | |
| Xcd gas detector - 5 nos | |
| Xcd gas detector - 7 nos | |
| Aerosol project lan installation charges | |
| Dell server | |
| Lan & net - gf stability chamber area modification | |
| Lan & network syst - ff admin area furniture work | |
| Lan & network system - first floor new stabiltiy c | |
| Lto-7 back up device | |
| Modular datacenter infrastructure smart | |
| Netbackup ent server wls 1 server hardwa | |
| Openscape business v2 x8 system | |
| Rack server, dual intel® xeon® e5-2620v4,rack serv | |
| Server & network system | |
| Storage emc vne 3200 | |
| Double Leaf Metal Door | |







Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadodara 390 003, Gujarat, India.

CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: alear@aleorderm.com | www.aleorderm.com



ii) Located at R&D facility of Aleor Dermaceuticals Limited at Survey No.: 342 of ALEAP Industrial Estate situated in Gajularamaram village, Qutbullapur Mandal, Medchal – Malkajgiri district - 500 090, Telangana:

| 0.8tr compact cassette indoor unit - 1 no's 0.8tr high wall split indoor unit - 2 no's 1.0 tr 4way cassette type indoor unit - 7 no's 1.7 tr 4way cassette type indoor unit - 7 no's 1000l stability chamber - 1 no's 100kvar apfc panel - 1 no's 100kg cream/ointment mfg plant - 1 no's 14 hp out door unit - 1 no's 14 hp out door unit - 1 no's 14 hp monoblock pump 2.0 tr 4way cassette type indoor unit - 2 no's 2.0tr high wall split indoor unit - 2 no's 2.0tr high wall split indoor unit - 4 no's 2.0tr aparallel redudant ups - 2 no's 2.4 hp out door unit - 1 no's 2.5 tr with 3000 cfm @ esp 100mm of wg(tfa) - 1 no' 2.8 hp out door unit - 2 no's 3.2 tr 4way cassette type indoor unit - 12 no's 3.0 hp out door unit - 2 no's 3.0 kva parallel redudant ups - 2 no's 400 kva dg - 1 no's 500 kva outdoor unitized substation - 1 no's 6 kva online double conversion ups - 1 no's 6 kva online double conversion ups - 1 no's 6 kva online double conversion ups - 1 no's 6 kva online double conversion ups - 1 no's 6 houndifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1 no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1 no's Fire hydrent system Hdpe tank 5000 liters | Asset Description | |
|--|--|------------|
| 0.8tr high wall split indoor unit - 2 no's 1.0 tr 4way cassette type indoor unit - 7 no's 1.7 tr 4way cassette type indoor unit - 7 no's 1000l stability chamber - 1no's 100kvar apfc panel - 1no's 10kg cream/ointment mfg plant - 1 no's 14 hp out door unit - 1 no's 14 hp out door unit - 1 no's 1hp monoblock pump 2.0 tr 4way cassette type indoor unit - 2 no's 2.0tr high wall split indoor unit - 2 no's 2.0tr high wall split indoor unit - 4 no's 2.0kva parallel redudant ups - 2no's 2.4 hp out door unit - 1 no's 2.5 tr with 3000 cfm @ esp 100mm of wg(tfa) - 1 no' 2.8 hp out door unit - 2 no's 3.2 tr 4way cassette type indoor unit - 12 no's 3.0 hp out door unit - 2 no's 3.0 kva parallel redudant ups - 2no's 400 kva dg - 1no's 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Fire alaram safety system Hdpe tank 5000 liters | 0.6tr compact cassette type indoor unit - 22 no's | |
| 1.0 tr 4way cassette type indoor unit - 7 no's 1.7 tr 4way cassette type indoor unit - 7 no's 10001 stability chamber - 1no's 100kvar apfc panel - 1no's 10ky cream/ointment mfg plant - 1 no's 14 hp out door unit - 1 no's 15 hp monoblock pump 2.0 tr 4way cassette type indoor unit - 2 no's 2.0 tr high wall split indoor unit - 2no's 2.0 tr high wall split indoor unit - 4 no's 2.0 tr way cassette type indoor unit - 4 no's 2.0 tr way cassette type indoor unit - 4 no's 2.0 tr way cassette type indoor unit - 4 no's 2.0 tr way cassette type indoor unit - 1 no's 2.1 tr with 3000 cfm @ esp 100mm of wg(tfa) - 1 no' 2.2 hp out door unit - 2 no's 3.2 tr 4way cassette type indoor unit - 12 no's 3.0 hp out door unit - 2 no's 3.0 hp out door unit - 2 no's 3.0 kva parallel redudant ups - 2no's 400 kva dg - 1no's 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static - 1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | 0.8tr compact cassette indoor unit - 1 no's | Ulista est |
| 1.7 tr 4way cassette type indoor unit - 7 no's 10001 stability chamber - 1no's 100kvar apfc panel - 1no's 10kg cream/ointment mfg plant - 1 no's 14 hp out door unit - 1 no's 1hp monoblock pump 2.0 tr 4way cassette type indoor unit - 2 no's 2.0tr high wall split indoor unit - 2no's 2.8 tr 4way cassette type indoor unit - 4 no's 20kva parallel redudant ups - 2no's 24 hp out door unit - 1 no's 25 tr with 3000 cfm @ esp 100mm of wg(tfa) - 1 no' 28 hp out door unit - 2 no's 3.2 tr 4way cassette type indoor unit - 12 no's 30 hp out door unit - 2 no's 30 hp out door unit - 2 no's 30kva parallel redudant ups - 2no's 400 kva dg - 1no's 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | 0.8tr high wall split indoor unit - 2 no's | |
| 1000l stability chamber - 1no's 100kvar apfc panel - 1no's 10kg cream/ointment mfg plant - 1 no's 14 hp out door unit - 1 no's 1hp monoblock pump 2.0 tr 4way cassette type indoor unit - 2 no's 2.0tr high wall split indoor unit - 2no's 2.8 tr 4way cassette type indoor unit - 4 no's 20kva parallel redudant ups - 2no's 24 hp out door unit - 1 no's 25 tr with 3000 cfm @ esp 100mm of wg(tfa) - 1 no' 28 hp out door unit - 2 no's 3.2 tr 4way cassette type indoor unit - 12 no's 3.0 hp out door unit - 2 no's 30 hp out door unit - 2 no's 30kva parallel redudant ups - 2no's 400 kva dg - 1no's 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | 1.0 tr 4way cassette type indoor unit - 7 no's | |
| 100kvar apfc panel - 1no's 10kg cream/ointment mfg plant - 1 no's 14 hp out door unit - 1 no's 1hp monoblock pump 2.0 tr 4way cassette type indoor unit - 2 no's 2.0tr high wall split indoor unit - 2no's 2.0tr high wall split indoor unit - 4 no's 2.0kva parallel redudant ups - 2no's 24 hp out door unit - 1 no's 25 tr with 3000 cfm @ esp 100mm of wg(tfa) - 1 no' 28 hp out door unit - 2 no's 3.2 tr 4way cassette type indoor unit - 12 no's 3.0 hp out door unit - 2 no's 3.0kva parallel redudant ups - 2no's 400 kva dg - 1no's 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | 1.7 tr 4way cassette type indoor unit - 7 no's | |
| 10kg cream/ointment mfg plant - 1 no's 14 hp out door unit - 1 no's 1hp monoblock pump 2.0 tr 4way cassette type indoor unit - 2 no's 2.0tr high wall split indoor unit - 2no's 2.8 tr 4way cassette type indoor unit - 4 no's 2.0kva parallel redudant ups - 2no's 2.4 hp out door unit - 1 no's 2.5 tr with 3000 cfm @ esp 100mm of wg(tfa) - 1 no' 2.8 hp out door unit - 2 no's 3.2 tr 4way cassette type indoor unit - 12 no's 3.0 hp out door unit - 2 no's 3.0kva parallel redudant ups - 2no's 400 kva dg - 1no's 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system Hdpe tank 5000 liters | | |
| 14 hp out door unit - 1 no's 1hp monoblock pump 2.0 tr 4way cassette type indoor unit - 2 no's 2.0tr high wall split indoor unit - 2no's 2.8 tr 4way cassette type indoor unit - 4 no's 2.0kva parallel redudant ups - 2no's 2.4 hp out door unit - 1 no's 2.5 tr with 3000 cfm @ esp 100mm of wg(tfa) - 1 no' 2.8 hp out door unit - 2 no's 3.2 tr 4way cassette type indoor unit - 12 no's 3.0 hp out door unit - 2 no's 3.0kva parallel redudant ups - 2no's 4.00 kva dg - 1no's 5.00 kva outdoor unitized substation - 1no's 6. kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | | |
| 1hp monoblock pump 2.0 tr 4way cassette type indoor unit - 2 no's 2.0 tr high wall split indoor unit - 2no's 2.8 tr 4way cassette type indoor unit - 4 no's 20kva parallel redudant ups - 2no's 24 hp out door unit - 1 no's 25 tr with 3000 cfm @ esp 100mm of wg(tfa) - 1 no' 28 hp out door unit - 2 no's 3.2 tr 4way cassette type indoor unit - 12 no's 30 hp out door unit - 2 no's 30kva parallel redudant ups - 2no's 400 kva dg - 1no's 50kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | 10kg cream/ointment mfg plant - 1 no's | |
| 2.0 tr 4way cassette type indoor unit - 2 no's 2.0tr high wall split indoor unit - 2no's 2.8 tr 4way cassette type indoor unit - 4 no's 20kva parallel redudant ups - 2no's 24 hp out door unit - 1 no's 25 tr with 3000 cfm @ esp 100mm of wg(tfa) - 1 no' 28 hp out door unit - 2 no's 3.2 tr 4way cassette type indoor unit - 12 no's 30 hp out door unit - 2 no's 30kva parallel redudant ups - 2no's 400 kva dg - 1no's 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | 14 hp out door unit - 1 no's | |
| 2.0tr high wall split indoor unit - 2no's 2.8 tr 4way cassette type indoor unit - 4 no's 20kva parallel redudant ups - 2no's 24 hp out door unit - 1 no's 25 tr with 3000 cfm @ esp 100mm of wg(tfa) - 1 no' 28 hp out door unit - 2 no's 3.2 tr 4way cassette type indoor unit - 12 no's 30 hp out door unit - 2 no's 30kva parallel redudant ups - 2no's 400 kva dg - 1no's 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | 1hp monoblock pump | |
| 2.8 tr 4way cassette type indoor unit - 4 no's 20kva parallel redudant ups - 2no's 24 hp out door unit - 1 no's 25 tr with 3000 cfm @ esp 100mm of wg(tfa) - 1 no' 28 hp out door unit - 2 no's 3.2 tr 4way cassette type indoor unit - 12 no's 30 hp out door unit - 2 no's 30kva parallel redudant ups - 2no's 400 kva dg - 1no's 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | 2.0 tr 4way cassette type indoor unit - 2 no's | |
| 20kva parallel redudant ups - 2no's 24 hp out door unit - 1 no's 25 tr with 3000 cfm @ esp 100mm of wg(tfa) - 1 no' 28 hp out door unit - 2 no's 3.2 tr 4way cassette type indoor unit - 12 no's 30 hp out door unit - 2 no's 30kva parallel redudant ups - 2no's 400 kva dg - 1no's 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system Hdpe tank 5000 liters | | |
| 24 hp out door unit - 1 no's 25 tr with 3000 cfm @ esp 100mm of wg(tfa) - 1 no' 28 hp out door unit - 2 no's 3.2 tr 4way cassette type indoor unit - 12 no's 30 hp out door unit - 2 no's 30kva parallel redudant ups - 2no's 400 kva dg - 1no's 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | 2.8 tr 4way cassette type indoor unit - 4 no's | |
| 25 tr with 3000 cfm @ esp 100mm of wg(tfa) - 1 no' 28 hp out door unit - 2 no's 3.2 tr 4way cassette type indoor unit - 12 no's 30 hp out door unit - 2 no's 30kva parallel redudant ups - 2no's 400 kva dg - 1no's 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | | |
| 28 hp out door unit - 2 no's 3.2 tr 4way cassette type indoor unit - 12 no's 30 hp out door unit - 2 no's 30kva parallel redudant ups - 2no's 400 kva dg - 1no's 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system Hdpe tank 5000 liters | 24 hp out door unit - 1 no's | |
| 3.2 tr 4way cassette type indoor unit - 12 no's 30 hp out door unit - 2 no's 30kva parallel redudant ups - 2no's 400 kva dg - 1no's 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | 25 tr with 3000 cfm @ esp 100mm of wg(tfa) - 1 no' | |
| 30 hp out door unit - 2 no's 30kva parallel redudant ups - 2no's 400 kva dg - 1no's 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | 28 hp out door unit - 2 no's | |
| 30kva parallel redudant ups - 2no's 400 kva dg - 1no's 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | | |
| 400 kva dg - 1no's 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | 30 hp out door unit - 2 no's | |
| 500 kva outdoor unitized substation - 1no's 6 kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | | |
| S kva online double conversion ups - 1no's Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | 400 kva dg - 1no's | |
| Air compressor Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | 500 kva outdoor unitized substation - 1no's | |
| Cabinet shower model: unicare ucss-52 Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | 6 kva online double conversion ups - 1no's | |
| Cental controller / bms integrator vrv - 1no's Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | Air compressor | |
| Dehumidifier Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | Cabinet shower model: unicare ucss-52 | |
| Electrical fittings and fixtures Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | Cental controller / bms integrator vrv - Ino's | |
| Exhaust centrifugal blower 1200cfm - 1no's Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | Dehumidifier | |
| Exhaust unit 2000cfm @ 25mm static -1 no's Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | Electrical fittings and fixtures | |
| Fire alaram safety system - 1no's Fire hydrent system Hdpe tank 5000 liters | Exhaust centrifugal blower 1200cfm - 1no's | |
| Fire hydrent system Hdpe tank 5000 liters | Exhaust unit 2000cfm @ 25mm static -1 no's | |
| Hdpe tank 5000 liters | Fire alaram safety system - 1no's | |
| | Fire hydrent system | |
| Hooter and annunciator system 16 channel - 1no | Hdpe tank 5000 liters | |
| | Hooter and annunciator system 16 channel - 1no | * |





Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vododara 390 003, Gujarat, India.

CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: aleor@aleorderm.com | www.aleorderm.com





| Asset Description |
|---|
| Hydrogen flameproof gas detector |
| Main lt panel -1 no's |
| Oxygen gas detector sensor |
| Ro water plant - 1no |
| Supply unit 150cfm @ 50mm static - 1 no's |
| Ups db -1no's |
| Walk-in stability chamber 25/60 - 1no |
| Walk-in stability chamber 30/65 -1no |
| Walk-in stability chamber 40/75 - 1no |
| 42u closed rack 800x800 - Ino's |
| C9300l 24p data networkswitch - 2no's |
| Catalyst 9200l 24-port poe+ - 1no's |
| Catalyst 9200l 48-port data - 3no's |
| Cisco aironet acess point - 10no's |
| Dell emc switch s4112t - 1no's |
| Dell poweredge r640 servers - 1no's |
| Expansion pair of mic for ss2 ex - 2no's |
| Electrical installation & equipments |

For Aleor Dermaceuticals Limited

Server & network system

Mitanshu Shah Director







Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Yadodara 390 003, Gujarat, India.
CIN: UZ4302GJ2016PLC092131 | PAN No :AAOCA44791 | Tel.: 0265 2280550 | E-mail: aleor@aleorderm.com | www.aleorderm.com



Registration with Various Authorities under respective laws, Bodies etc.:

i) Registrations:

| Sr. No. | Name of Authority | Nature of Registration/ License | Registration/License No. | | |
|------------|--|---|----------------------------------|--|--|
| 1 | Income Tax Department | PAN | AAOCA4479J | | |
| 2 | Income Tax Department | TAN | BRDA05174B | | |
| 3 | Registrar of Company's CIN | Registration No. | U24302GJ2016PLC09 2131 | | |
| 4 | PF Registration No. | Registration No. | VDBRD1504425 | | |
| 5 | ESI Registration | Registration No. | 52380435780010305 (Hyderabad) | | |
| 6 | ESI Registration | Registration No. | 38380435780010305 (Vadodara) | | |
| 7 | Factory Licence No. | License No. | 34507 | | |
| 8 | Professional Tax Registration No. | Registration No. | PRC021003663 | | |
| 9 | HDFC Life Gratuity Policy No. | Policy No. | G0001591 | | |
| 10 | Employee's Deposit Linked Insurance Scheme | Policy No. | VDBRD1504425 | | |
| 11 | Contractual Labour Registration No. | Registration No. | CLRA/681/1683/2015 | | |
| 12 | Central Board of Indirect Taxes and Customs | Goods And Service Tax, State – Gujarat | 24AAOCA4479J1Z8 | | |
| 13 | Central Board of Indirect Taxes and Customs | Goods And Service Tax, State – Telangana | 36AAOCA4479J1Z3 | | |
| 14 | Central Board of Indirect Taxes and Customs | Letter Of Undertaking, State – Gujarat | AD2403220180270 | | |





Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadodara 390 003, Gujarat, India.

CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: alear@alearderm.com | www.alearderm.com



| A | A THE CO |
|----|--------------|
| A | LEOR |
| S+ | Name of Auth |

| Sr. No. | Name of Authority | Nature of Registration/ License | Registration/License No. | | |
|------------|---|---|--|--|--|
| 15 | Central Board of Indirect Taxes and Customs | Letter Of Undertaking, State – Telangana | AD3603220184990 SELF-SEALING PERMISSION NO: 160/CUS/AHD/2019- | | |
| 16 | Office of the Principal Commissioner Of Customs, Custom House | Self-Sealing Factory Stuffing Permission, State Gujarat | | | |
| 17 | Central Board of Excise & Customs | | | | |
| 18 | Central Board of Excise & Customs | Pre GST Registration - Service Tax Registration | AAOCA4479JSD001 | | |
| 19 | Jt. Directorate General of Foreign Trade, Vadodara Directorate General of Foreign Trade, New Delhi | One Star Export House | VDDSTATAPPLY00000 005AM23 DTD. 24-05- 2022 VALID UPTO 24- 04-2027 | | |
| 20 | Jt. Directorate General of Foreign Trade, Vadodara Directorate General of Foreign Trade, New Delhi | IEC Certificate | 3416902521 DT. 08- 07-2016 | | |
| 21 | Pharmaceuticals Export Promotion Council, Hydrerabad, Ahmedabad | Pharmexcil (Manufacturer Exporter) | PXL/LSM/8182/2016- 17 VALID UPTO 31-03- 2023 | | |
| 22 | Jyoti Financial Services | Digital Signature Key For Online Application | G. SHERLY (VALID UPTO 28-01-2023 & 21-10-2023 | | |
| 23 | Town Planning & Development, Vadodara | Land Approval | 77,97,76,790 | | |
| 24 | Wireless Planning and Co ordanating wing, Mumbai | Wireless Walky Talky | 3291/WRLO-19/1-11 | | |







Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadodara 390 003, Gujarat, India.
CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: alear@alearderm.com | www.alearderm.com



| A | THE WAY |
|----|--------------|
| K | IFOR |
| ST | Name of Auth |

| Sr. No. | Name of Authority | Nature of Registration/ License | Registration/License No. |
|------------|---|---|------------------------------|
| 25 | Petroleum & Explosive Safety Organization (PESO), Vadodara & Nagpur. | Petroleum Class A Storage " Bulk Alcohol Storage " 18 KL | P/WC/GJ/16/442 (P401317) |
| | | Petroleum Class B Storage "Bulk HSD Storage "30 KL | P/HQ/GJ/15/5506 (P401958) |
| | | Import of Gas Cylinders | G/HQ/GJ/04/71(G65 159) |
| | | Import of Gas Cylinders | G/HQ/GJ/04/304 (G73631) |
| | | Import of Gas Cylinders | G/HQ/GJ/04/533 (G84857) |
| | | Import of Gas Cylinders | G/HQ/GJ/04/850(G1 00936) |
| | á | License to store compressed gas in cylinders | G/WB/GJ/06/520 (G60556) |
| 25 | Chief Inspector of Lift and Escalators, | Lift License (Admin Area) | G/SZ/19/8147/2018 |
| | Vadodara. | Lift License (Immono Area) Nr. South Canteen | G/SZ/19/8066/2018 |
| | | Lift License (Immono Area) Nr. South Canteen | G/SZ/19/8065/2018 |
| | | Lift License (QC Area) | G/SZ/19/10042/2019 |
| 27 | Gujarat State Fire Prevention Services, Vadodara. | Fire NOC | 69/530/2020 |









Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vododara 390 003, Gujarat, India.

CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: aleor@aleorderm.com | www.aleorderm.com



| Sr. No. | Name of Authority | Nature of Registration/ License | Registration/License No. |
|------------|---|--|---|
| 28 | GPCB, Gandhinagar. | Consent to Establishment | NOC-VRD-4062/ID- 52788/550448 CTE No.105911 |
| | | Consent & Authorisation | A-98789 |
| | | Bio Medical Waste authorization | BMW-342692 |
| 29 | Central Ground Water Authority, Delhi. | NOC For Ground Water Abstraction | CGWA/NOC/IND/ORI G/2019/6685 |
| 30 | MGVCL/Central Bank Of India.Vadodara | Bank Guaranty for Electrical HT Connection | 009BG01210490001 |
| 31 | Office of Electrical Inspector, Vadodara | Initial Inspection Installation of 2x2000kva transformer along with HT Breaker panel and Associated electrical Equipment | EI/VDR/INSP/4223/1 7 |
| 32 | Office of Electrical Inspector, Vadodara | Initial Inspection Installation of 2x1500kva DG Set to work as a standby source | EI/VDR/INSP/6584/1 7 |
| 33 | MGVCL, Vadodara. | MGVCL 1500 KVA HT Power Released | RO:2499 |
| 34 | Office of the Collector of Electricity Duty, Vadodara | Registration of Generating set/s under Gujarat Electricity Duty Act,1958 (DG-1) | BRD/000004-0402018 |



Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vododara 390 003, Gujarat, India.

CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: aleor@aleorderm.com





| Sr. No. | Name of Authority | Nature of Registration/ License | Registration/License No. | |
|--|--|--|-----------------------------|--|
| 35 Office of the Collector of Electricity Duty, Vadodara | | Registration of Generating set/s under Gujarat Electricity Duty Act, 1958 (DG-2) | BRD/000003-0303018 | |
| 35 | MGVCL, Vadodara | Exemption from Electrical Duty | MGVCL/JMB/REV.20 /2823 | |
| 37 | Ministry of Commerce. | Industrial Enterpreneur Memorandum | 2095/5IA/IMO/2016 | |
| 38 | DGFT | GS-I | 8908012115 | |
| 39 | Prohibition & Excise, Vadodara | Nokarnama - Nikhil Vispute - RS-II & SA-I | 204/2018-19 | |
| | | Nokarnama - Manish Kumar Pathak - RS-II & SA-I | 244/2017-18 | |
| | | Nokarnama - Manish Kumar Pathak - AC-II | 169/2019-20 | |
| | | Nokarnama - Prasoon Gupta - AC-II | 170/2019-20 | |
| 40 | Ministry of Micro, Small & Medium Enterprises (MSME) (Regular Basis) | Annual Production Programme Tavaborole Topical Solution 5% Efinaconazole Topical Solution | 1386683 | |









Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadodara 390 003, Gujarat, India.
CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: aleor@aleorderm.com | www.oleorderm.com



| Sr. No. | Name of Authority | Nature of Registration/ License | Registration/License No. T.NO-442 1661752 | |
|------------|--|---|---|--|
| 41 | Ministry of Micro, Small & Medium Enterprises (MSME) (Regular Basis) | Annual Production Programme Clobetasol Propionate Shampoo 0.05% Diclofenac Sodium Topical Solution 2% | | |
| 42 | Ministry of Micro, Small & Medium Enterprises (MSME) (Regular Basis) | Annual Production Programme Clobetasol Propionate Spray 0.05% | | |
| 43 | Ministry of Micro, Small & Medium Enterprises (MSME) | Annual Production Programme Clobetasol Propionate Foam 0.05% Betamethasone Valerate Foam 0.12% | 1610266 | |
| 44 | Ministry of Micro, Small & Medium Enterprises (MSME) | Annual Production Programme Minoxidil | . | |
| 45 | FDCA - 20B, 21B | Licence to Sale, Stock or Exhibit or offer for sale, distribute | GJ-VAD-160880/81 | |
| 46 | Pharmaceutical Export Promotion Council | Pharmexcil | PXL/HO/BCEXMP/USA /54/61/62/217 | |



Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadodara 390 003, Gujarat, India.

CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: aleor@aleorderm.com | www.aleorderm.com



| Sr. No. | Name of Authority | Nature of Registration/ License | Registration/License No. | |
|------------|--|---------------------------------------|---|--|
| 47 | Office of Controller - Legal Metrology | Bullions Weight | 2329544/VAD/2022/01 | |
| 48 | Office of Controller - Legal Metrology | Weighing Balance | 2345001/VAD/2022/0 | |
| 49 | FDCA, Gandhinagar | WHO GMP | G/25/2216 in Form 25 | |
| 50 | FDCA, Gandhinagar | WHO GMP | G/25/2216 in Form 28 | |
| 51 | Government of Telangana | Building Plan approval | 45222 | |
| 52 | Government of Telangana | Consent Order | D.DIS.NO.A/MDCL- I/38/2021 dated 22.02.2021 | |
| 53 | TSPCB, Hyderabad | Fire NOC | No:1705-RR- II/TSPCB/ZOH/TS- iPASS/CFO/2021-226 dated 17.07.2021 | |
| 54 | TSPCB, Hyderabad | Lift | | |
| 55 | Drugs Control Administration Government of Telangana | FORM -29 | TS/MDL/2021-79838 | |

ii) Advance Licenses / Advance Authorizations:

| Name of | Registration No. | | | | |
|-------------------------------|------------------|------------|----------------|--------------------------------------|--|
| Authority | Sr. | License | | File No. | |
| | No. | No. | Date | | |
| Jt. Directorate General of | 1 | 3410044950 | 03-05- 2019 | 34/24/040/00370/AM19 (EODC recd.) | |
| Foreign Trade, Vadodara | 2 | 3410044985 | 19-03- 2019 | 34/24/040/00396/AM19 (EODC recd.) | |
| Directorate | 3 | 3410044986 | 19-03- 2019 | 34/24/040/00397/AM19 (EODC recd.) | |
| General of Foreign Trade, | 4 | 3410045318 | 26-07- 2019 | 34/24/040/00136/AM20 | |
| New Delhi | 5 | 3410045317 | 26-07- 2019 | 34/24/040/00135/AM20 | |





Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadadara 390 003, Gujarat, India.

CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: aleor@aleorderm.com | www.aleorderm.com





| Name of | Registration No. | | | | | |
|-------------------------------|------------------|------------|----------------|----------------------|--|--|
| Authority | Sr. | Licens | e | File No. | | |
| | No. | No. | Date | | | |
| Jt, Directorate General of | 6 | 3410045849 | 02-06- 2020 | 34/24/040/00339/AM20 | | |
| Foreign Trade, Vadodara | 7 | 3410045850 | 02-06- 2020 | 34/24/040/00342/AM20 | | |
| Directorate | 8 | 3410046235 | 22-07- 2020 | 34/24/040/00098/AM21 | | |
| General of Foreign Trade, | 9 | 3410046293 | 17-08- 2020 | 34/21/040/00160/AM21 | | |
| New Delhi | 10 | 3410046419 | 25-09- 2020 | 34/24/040/00169/AM21 | | |
| | 11 | 3410046457 | 10-09- 2020 | 34/21/040/00229/AM21 | | |
| | 12 | 3410046497 | 28-10- 2020 | 34/24/040/00205/AM21 | | |
| | 13 | 3410046546 | 13-11- 2020 | 34/21/040/00320/AM21 | | |
| | 14 | 3410046549 | 13-11- 2020 | 34/24/040/00223/AM21 | | |
| | 15 | 3411000264 | 05-03- 2021 | 34AA04002785AM21 | | |
| | 16 | 3411000366 | 30-03- 2021 | 34AA04003406AM21 | | |
| | 17 | 3411000565 | 11-05- 2021 | 34AA04001263AM22 | | |
| | 18 | 3411000721 | 06-11- 2021 | 34AA04002310AM22 | | |
| | 19 | 3411000813 | 07-07- 2021 | 34AA04004084AM22 | | |
| | 20 | 3411000938 | 04-08- 2021 | 34AA04004194AM22 | | |
| | 21 | 3411001056 | 02-09- 2021 | 34AA04005756AM22 | | |
| | 22 | 3411001110 | 15-09- 2021 | 34AA04007315AM22 | | |
| | 23 | 3411001396 | 15-11- 2021 | 34AA04009242AM22 | | |



Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadodara 390 003, Gujarat, India.

CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: aleor@aleorderm.com | www.aleorderm.com





| Name of | Registration No. | | | | |
|-------------------------------|------------------|------------|----------------|------------------|--|
| Authority | Sr. License | | e | File No. | |
| gG | No. | No. | Date | | |
| Jt. Directorate General of | 24 | 3411001409 | 17-11- 2021 | 34AA04009240AM22 | |
| Foreign Trade, Vadodara | 25 | 3411001429 | 22-11- 2021 | 34AA04009399AM22 | |
| Directorate | 26 | 3411001537 | 14-12- 2021 | 34AA04009400AM22 | |
| General of Foreign Trade, | 27 | 3411001739 | 28-01- 2022 | 34AX04000203AM22 | |
| New Delhi | 28 | 3411001937 | 09-03- 2022 | 34AX04000347AM22 | |
| | 29 | 3411002246 | 13-05- 2022 | 34AX04000127AM23 | |
| IV. | 30 | 3411002514 | 08-07- 2022 | 34AB04000647AM23 | |
| | 31 | 3411002535 | 15-07- 2022 | 34AX04000361AM23 | |

iii) MEIS Scrip:

| Name of | Registration No. | | | | | |
|---|------------------|-------------------|------------------------|--------------------------------|--|--|
| Authority | Sr. | License | | File No. | | |
| | No. | No. 3419024476 | Date 26-10- 2021 | 34/21/090/51955/AM22 (Sold) | | |
| Jt. Directorate General of | 1 | | | | | |
| Foreign Trade, Vadodara | 2 | 3419022271 | 26-10- 2021 | 34/21/090/51950/AM22 (Sold) | | |
| Directorate | 3 | 3419024454 | 26-10- 2021 | 34/21/090/51933/AM22 (Sold) | | |
| General of Foreign Trade, New Delhi | 4 | 3419025260 | 21-12- 2021 | 34/21/090/52743/AM22 (Sold) | | |







Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadodara 390 003, Gujarat, India.

CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: aleor@aleorderm.com | www.aleorderm.com





EPCG License (Export Promotion for Capital Goods):

| Name of | Registration No. | | | | | |
|-------------------------------|------------------|------------|----------------|----------------------|--|--|
| Authority | Sr. | License | | File No. | | |
| | No. | No. | Date | | | |
| Jt. Directorate General of | 1 | 3430003044 | 30-03- 2017 | 34/21/021/00205/AM17 | | |
| Foreign Trade, Vadodara | 2 | 3430003045 | 30-03- 2017 | 34/21/021/00206/AM17 | | |
| Directorate | 3 | 3430003065 | 08-05- 2017 | 34/21/021/00009/AM18 | | |
| General of Foreign Trade, | 4 | 3430003058 | 25-04- 2017 | 34/21/021/00008/AM18 | | |
| New Delhi | 5 | 3430003059 | 27-04- 2017 | 34/21/021/00011/AM18 | | |
| 2 | б | 3431000105 | 15-01- 2021 | 34EAEPC01923AM21 | | |
| | 7 | 3431002267 | 18-05- 2022 | 34EAEPC00050AM23 | | |

v) Prohibition and Excise:

| Name of | Nature of | Registration No. | | | | | |
|--|----------------------------------|------------------|---------------------------|------------------------|------------------|--|--|
| Authority | Registration | Sr. No. | License No. | Renewed/ Issued Dt. | Renewed Up to | | |
| Prohibition & Excise - | R.SII | 1 | 283/22-23 | 05-06- 2022 | 31-03- 2023 | | |
| Vadodara & Gandhinagar | D.SV | 2 | S-111/2022- 23 | 22-06- 2022 | 31-03- 2023 | | |
| | S.AI | 3 | 04/2022-23 | 06-01- 2022 | 31-03- 2023 | | |
| | A.CII | 4 | 225/2022-23 | 30-03- 2022 | 31-03- 2023 | | |
| Office of District Prohibition & | RECTIFIED SPIRIT (ETHANOL) | 1 | Cr.No.E3/12 0/2022 | 14-03- 2022 | 31-03- 2023 | | |
| Excise Office Medchal - Malkajgiri District | METHANOL | 2 | Procgs.No.E3 /116/2022 | 14-03- 2022 | 31-03- 2023 | | |



Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadodara 390 003, Gujaret, India.

CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: alear@aleorderm.com | www.aleorderm.com





| Name of | Nature of | Registration No. | | | | | |
|--|---|------------------|-----------------|------------------------|------------------|--|--|
| Authority | Registration | Sr. No. | License No. | Renewed/ Issued Dt. | Renewed Up to | | |
| Narcotics Control Bureau, Ahmedabad | Registration No. To Procure, Possession, Consumption of Acetic Anhdride | 1 | AHCD01009 22 | 21-03- 2018 | NA | | |

vii) Annual Production Programme - Industries Commissionerate Office:

| Name of Authority | Nature of Registration | Sr. No. | APP No. | APP Issued Date |
|---|---------------------------------|------------|------------------|--------------------|
| Industries Commissionerate | Spirit (Alcohol) | 1 | APP/2019-20/442 | 29-07- 2019 |
| office, Udyog Bhavan, Gandhinagar | Annual Production Program | 2 | APP/2020/145 | 09-05- 2020 |
| | | 3 | APP/2020/1661752 | 22-09- 2020 |
| | | 4 | APP/2022/261 | 15-02- 2022 |
| | | 5 | APP/2022/837 | 17-05- 2022 |
| | | 6 | APP/2022/1051 | 17-06- 2022 |



viii) Schedule related to Exports Duty Drawback (All Industry rate):

Duty Drawback (All industry claims) which are received or to be received, may be transferred to Alembic Pharmaceuticals Limited after the process of amalgamation.







Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadadara 390 003, Gujarat, India.

CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: alear@alearderm.com | www.alearderm.com



| Sr. No. | Name of the Authority | uthority | | Status | |
|------------|--|---|--------|----------------|--|
| 1 | US FDA | Clobetasol Propionate Spray, 0.05% | 211191 | Approved | |
| 2 | US FDA | Lidocaine Ointment USP, 5% | 211469 | Approved | |
| 3 | US FDA | Clobetasol Propionate Ointment USP, 0.05% | 211800 | Approved | |
| 4 | US FDA | Efinaconazole Topical Solution, 0% | 212027 | Tent. Approved | |
| 5 | US FDA | Tavaborole Topical Solution, 5% | 212188 | Approved | |
| 6 | US FDA | Desonide Ointment, 0.05% | 212473 | Approved | |
| 7 | US FDA | Diclofenae Sodium Topical Solution, 2% | 212506 | Tent. Approved | |
| 8 | US FDA | Clobetasol Propionate Topical Solution, 0.05% | 212881 | Approved | |
| 9 | US FDA | Testosterone Topical Solution, 30 mg per 1.5 mL | 212882 | Approved | |
| 10 | US FDA | Clobetasol Propionate Cream, 0.05% | 213291 | Approved | |
| 11 | US FDA | Clobetasol Propionate Shampoo, 0.05% | 213290 | Approved | |
| 12 | US FDA | Adapalene Gel USP, 0.3% | 213508 | Approved | |
| 13 | US FDA | Acyclovir Cream, 5% | 212361 | Filed | |
| 14 | US FDA | Desonide Lotion, 0.05% | 213632 | Approved | |
| 15 | US FDA | Lidocaine and Prilocaine Cream USP, 2.5%/2.5% | 213923 | Approved | |
| 16 | US FDA | Dapsone Gel, 5% | 213414 | Filed | |
| 17 | US FDA Diclofenac Sodium Topical Gel, 3% | | 212351 | Filed | |



Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadadara 390 003, Gujarat, India.
CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: alean@aleanderm.com | www.aleanderm.com





| Sr. Name of the No. Authority | | Authority | | Status | |
|----------------------------------|--------------------------------------|--|---------|---------------|--|
| 18 | US FDA | Testosterone Gel, 1.62% | | Approved | |
| 19 | US FDA | Metronidazole Gel USP, 1% | 212646 | Approved | |
| 20 | US FDA | Nystatin & Triamcinolone Acetonide Cream 100,000 Units/gm & 0.1% | | Approved | |
| 21 | US FDA | Mupirocin Cream USP, 2% | 213053 | Approved | |
| 22 | US FDA | Nystatin & Triamcinolone Acetonide Ointment 100,000 Units/gm & 0.1% | 214186 | Approved | |
| 23 | US FDA . | Clindamycin Phosphate Topical Gel USP, 1% | 212466 | Filed | |
| 24 | US FDA | Penciclovir Cream, 1% | 212352 | Filed | |
| 25 | US FDA | Desonide Cream, 0.05% | 214396 | Filed | |
| 26 | US FDA | Adapalene and Benzoyl Peroxide Topical Gel, 0.3%/2.5% | 214185 | Filed | |
| 27 | US FDA | Docosanol Cream, 10% | 215839 | Approved | |
| 28 | US FDA | Betamethasone Valerate Foam, 0.12% | 215832 | Filed | |
| 29 | US FDA | Clobetasol Propionate Foam, 0.05% | 215838 | Approved | |
| 30 | US FDA | Dapsone Gel, 7.5% | 215718 | Filed | |
| 31 | Canada-For Sandoz Inc. | Efinaconazole Topical Solution 10 % w/w | e239614 | Tent.Approved | |
| 32 | Canada- For Pharmascience Inc. | Efinaconazole Topical Solution 10% w/w | e257476 | Filed | |







Mitanshu Shah Director



Aleor Dermaceuticals Ltd

Registered Office: 5th Floor, Administrative Building, Alembic Limited, Alembic Road, Vadodaro 390 003, Gujarat, India. CIN: U24302GJ2016PLC092131 | PAN No :AAOCA4479J | Tel.: 0265 2280550 | E-mail: aleor@aleorderm.com | www.aleorderm.com

| Prepared by | l'mal |
|-------------|-------|
| | ap |
| Signature | 1 |

Date 8 30- 8 2022

